# Efficient Computation of State-Constrained Reachability Problems Using Hopf–Lax Formulae

Donggun Lee <sup>D</sup> and Claire J. Tomlin <sup>D</sup>, Fellow, IEEE

Abstract—This article considers two state-constrained reachability problems: computing 1) control-invariant and 2) reach-avoid sets, both under state constraints. Prior research has developed Hamilton-Jacobi (HJ) partial differential equations (PDEs) that characterize the optimal cost functions of these two problems. Unfortunately, solving the HJ PDEs by grid-based methods, such as level-set methods, suffers from exponentially growing computational complexity in the system state dimension. In order to alleviate this computational issue, this article proposes a Hopf-Lax formula for each reachability problem's HJ PDE. The advantage of the Hopf-Lax formulae is that they have more favorable convexity conditions than the corresponding problems. Thus, direct methods may be used to solve Hopf-Lax formulae and thus efficiently compute the optimal solution of the reachability problems under specified conditions. This article provides an example for each reachability problem and demonstrates the performance of the proposed Hopf–Lax method.

IEEE CSS

*Index Terms*—Nonlinear control systems, optimal control, scalability.

## I. INTRODUCTION

T HIS article focuses on two reachability problems: 1) the state-constrained control-invariance problem (SCCIP) and 2) the state-constrained reach-avoid problem (SCRAP) [1], [2]. The SCCIP provides a quantitative measure for control-invariance under constraints. Mathematically, the problem aims to find a control signal to minimize the maximum distance from the target in a time interval while satisfying state constraints. To relate it to a well-known construct, the viability

Manuscript received 5 May 2022; revised 9 November 2022; accepted 15 January 2023. Date of publication 31 January 2023; date of current version 26 October 2023. This work was supported in part by DARPA through HICON-LEARN under Grant FA8750-18-C-0101, in part by SRC through CONIX under Grant 1042741-394321, and in part by ONR through BRC under Grant N00014-18-1-2214. Recommended by Associate Editor E. C. Kerrigan. (*Corresponding author: Donggun Lee.*)

Donggun Lee is with the Department of Mechanical Engineering, University of California, Berkeley, CA 94703 USA (e-mail: donggun\_ lee@berkeley.edu).

Claire J. Tomlin is with the Department of Electrical Engineering and Computer Sciences, University of California, Berkeley, CA 94703 USA (e-mail: tomlin@eecs.berkeley.edu).

Color versions of one or more figures in this article are available at https://doi.org/10.1109/TAC.2023.3241180.

Digital Object Identifier 10.1109/TAC.2023.3241180

kernel [3] is that set of initial states whose SCCIP quantity is less than a certain value. The SCRAP quantifies the distance between the system and a target while the system maintains safety constraints [2]. Mathematically, in this problem, we aim to find a control signal to minimize the minimum distance to the goal over a time interval while satisfying state constraints. For example, the reach–avoid set [4] is the set of initial states whose SCRAP quantity is less than a certain value.

Hamilton-Jacobi (HJ) frameworks build on dynamic programming and viscosity theory to solve optimal control problems [5], including SCCIP and SCRAP. Viscosity theory is a notion of weak solutions for first-order partial differential equations (PDEs) to deal with nondifferentiability of the solution. Various HJ methods have been developed for stateunconstrained problems [3], [6], [7], [8], [9], [10], [11] and for state-constrained problems [12], [13], [14], [15], [16], [17], [18], [19], [20], [21], [22]. State-constrained HJ PDEs [12], [13], [14], [15], [16], [17] develop viscosity theory under controllability assumptions, which allow the system to maintain a state constraint, which does not, unfortunately, hold for general systems. More recent state-constrained HJ work [18], [19], [20], [21], [22] does not need the controllability assumption because it utilizes an epigraphical technique. The epigraphical technique does not find an HJ PDE for the state-constrained problem's optimal cost. Instead, it finds an HJ PDE for another value function whose subzero-level set is the epigraph of the optimal cost of the state-constrained problem. Here, standard viscosity theory [18] can be used. For SCCIP and SCRAP, Lee and Tomlin [1] and Lee et al. [2] utilize the epigraphical technique to propose HJ PDEs.

The solution of HJ PDEs can be computed numerically using, for example, level-set or fast marching methods [23]. These require gridding the state space and thus suffer from computational complexity that grows exponentially in the system state dimension. There have been various methods to alleviate this computational issue, including optimization with approximation techniques [24], [25], [26], [27], control-barrier-function-based methods [28], [29], [30], [31], [32], geometry-based formulations [33], [34], [35], [36], [37], temporal logic [38], [39], learning-based approaches [40], [41], [42], [43], and Hopf–Lax theory [44].

© 2023 The Authors. This work is licensed under a Creative Commons Attribution 4.0 License. For more information, see http://creativecommons.org/licenses/by/4.0/ Given an HJ PDE, Hopf–Lax theory finds an optimization problem whose optimal cost is the viscosity solution to the HJ PDE [44], [45], [46], [47]; the optimization problem is called a Hopf–Lax formula. Hopf–Lax formulae can significantly reduce computation, as compared to the original HJ PDE, because they do not require gridding the state space [48], [49], [50], [51]. Hopf–Lax formulae can be solved by combinations of direct methods (multiple shooting [52] or collocation methods [53]) and numerical optimization methods, such as interior-point methods [54, Ch. 11], sequential quadratic programming [55, Ch. 18], or the split Bregman method [56]. As such, for some problems, the computational complexity of solving Hopf–Lax formulae can be polynomial in the state dimension [48].

State-of-the-art Hopf–Lax theory has computed viscosity solutions to HJ PDEs representing state-unconstrained problems [44], [46], [48], [49], [50], [57]. On the other hand, there is a small number of papers dealing with HJ PDEs for state-constrained problems. [22] deals with HJ PDEs whose Hamiltonian has only costate dependency. Lee et al. [51] and Lee and Tomlin [58] present Hopf–Lax formulae for a particular state-constrained problem for nonlinear systems. Lee and Tomlin [58] present the Hopf–Lax formula for the SCRAP in a time-invariant case: cost, dynamics, and state constraints are time-invariant. In this prior work on Hopf–Lax theory for HJ PDEs relevant to state-constrained problems, there has not been an analysis as to whether the computed results are viscosity solutions.

This article builds on HJ PDEs for the SCCIP and SCRAP problems, which have first been presented in [1] and [2]. The current article presents Hopf–Lax theory to compute the optimal cost of these problems whose epigraph is characterized by the viscosity solutions to the HJ PDEs. While the work in [58] first presents the Hopf–Lax formula for the time-invariant SCRAP, it does not present the viscosity solution analysis, and only considers a zero stage cost.

For each of the SCCIP and SCRAP, we consider two cases: 1) a time-varying case where cost, dynamics, and state constraints are time-varying; and 2) a time-invariant case where those are time-invariant [1], [2]. These two cases correspond to different Hopf–Lax formulae, resulting in different convexity conditions. Thus, we will choose one of the Hopf–Lax formulae that have more favorable convexity conditions and then apply direct methods to find the optimal cost. On the other hand, the Hopf–Lax formula assumes a convex Hamiltonian in the costate. Since the Hamiltonian in the HJ PDE for the time-invariant SCCIP is nonconvex, there is no Hopf–Lax formula for the time-invariant SCCIP. Thus, this article presents three Hopf–Lax formulae: one for the time-varying SCCIP, one for the time-varying SCRAP, and one for the time-invariant SCRAP.

In summary, our contribution is fourfold as follows.

- We propose three Hopf-Lax formulae for the timevarying SCCIP, for the time-varying SCRAP, and for the time-invariant SCRAP. Previously, Lee and Tomlin [58] first proposed the Hopf-Lax formula for the time-invariant SCRAP, whose stage cost is zero, and this article extends it for nonstage-cost problems.
- We present the first proof of viscosity solutions from Hopf–Lax theory relevant to state-constrained optimal control problems.

- 3) We identify conditions under which direct methods, such as multiple shooting or collocation methods, result in convex problems for the Hopf–Lax formulae but not for the original problems (SCCIP and SCRAP). This article also proves that if the direct methods result in convex problems for the original problems, their Hopf–Lax formulae always result in convex problems.
- We present examples for SCCIP and SCRAP to demonstrate the utility and performance of the proposed Hopf– Lax formulae.

The organization of this article is as follows. Section II defines both the SCCIP and SCRAP. Section III presents the HJ PDEs whose solutions' subzero-level sets are the corresponding epigraphs of the SCCIP or SCRAP. Section IV proposes our new theorem in viscosity theory, which is utilized to prove our main results, the Hopf–Lax formulae for the SCCIP, and SCRAP in Section V. Section VI presents convexity analysis for the SCCIP, SCRAP, and their Hopf–Lax formulae. Section VII presents two examples to demonstrate the utility and performance of the proposed Hopf–Lax formulae for each class of problem, and Section VIII concludes the article.

## A. Notation

This article uses the subscript \* to denote optimality and superscript \* to denote the Legendre–Fenchel transformation:  $\alpha_*$ denotes an optimal control signal, and  $H^*$  denotes the Legendre– Fenchel transformation of a function H.  $H^{**} = (H^*)^*$  denotes the biconjugate of a function H: the Legendre–Fenchel transformation of the Legendre–Fenchel transformation of a function H. "co(B)" denotes a convex-hull operator of a set B.

## II. STATE-CONSTRAINED REACHABILITY PROBLEMS AND THE HJ ANALYSIS

This section introduces the two state-constrained reachability problems: 1) the SCCIP and 2) the SCRAP. In these, we are considering the state trajectory  $(x : [t, T] \rightarrow \mathbb{R}^n)$  in a time interval [t, T] solving the following ordinary differential equation (ODE):

$$\dot{\mathbf{x}}(s) = f(s, \mathbf{x}(s), \alpha(s)), s \in [t, T], \text{ and } \mathbf{x}(t) = x \quad (1)$$

where (t, x) are the initial time and state, s is time between t and T,  $f : [t, T] \times \mathbb{R}^n \times A \to \mathbb{R}^n$  is a dynamics function,  $A \subset \mathbb{R}^m$  is the control set, and  $\alpha \in \mathcal{A}(t)$  is the measurable control signal

$$\mathcal{A}(t) := \{ \alpha : [t,T] \to A \mid \|\alpha\|_{L^{\infty}(t,T)} < \infty \}.$$
(2)

We assume that A is a compact subset in  $\mathbb{R}^m$ .

State-constrained control-invariance problem (SCCIP): For given initial time and state (t, x), solve

$$\vartheta_1(t,x) := \inf_{\alpha \in \mathcal{A}(t)} \max_{\tau \in [t,T]} \int_t^\tau L(s, \mathbf{x}(s), \alpha(s)) \mathrm{d}s + g(\tau, \mathbf{x}(\tau))$$
(3)

subject to 
$$c(s, \mathbf{x}(s)) \le 0, \quad s \in [t, T]$$
 (4)

where x solves (1),  $L : [t, T] \times \mathbb{R}^n \times A \to \mathbb{R}$  is the stage cost,  $g : \mathbb{R} \times \mathbb{R}^n \to \mathbb{R}$  is the terminal cost, f is the system dynamics as defined above,  $c : [t, T] \times \mathbb{R}^n \to \mathbb{R}$  is the state constraint, and  $\tau$  is the time when the cost is the highest. The scalar function c can handle a number of state constraints. For example, consider state constraints  $c_1, \ldots, c_l(s, \mathbf{x}(s)) \leq 0$ ;  $c = \max\{c_1, \ldots, c_l\}$  encodes all state constraints into a scalar constraint function. We would like to note that  $\vartheta_1(t, x) \in \mathbb{R} \cup \{\infty\}$ . This means that, for initial time and state (t, x), if there does not exist a control signal  $\alpha$  to satisfy the state constraint (4), the optimal cost  $\vartheta_1$  is  $\infty$ .

Assumption 1. (Lipschitz continuity and compactness):

- 1) The control set A is compact and convex.
- 2)  $f: [0,T] \times \mathbb{R}^n \times A \to \mathbb{R}^n$  (f = f(t, x, a)) is Lipschitz continuous in (t, x) for each  $a \in A$ .
- 3) The stage cost  $L: [0,T] \times \mathbb{R}^n \times \mathbb{R}^m \to \mathbb{R}$  (L = L(t,x,a)) is Lipschitz continuous in (t,x) for each  $a \in A$ .
- 4) For all  $(t,x) \in [0,T] \times \mathbb{R}^n$ ,  $\{f(t,x,a) \mid a \in A\}$  and  $\{L(t,x,a) \mid a \in A\}$  are compact and convex.
- 5) The terminal cost  $g: [0,T] \times \mathbb{R}^n \to \mathbb{R}$  (g = g(t,x)) is Lipschitz continuous in (t,x).
- 6) The state constraint  $c : [0, T] \times \mathbb{R}^n \to \mathbb{R}$  (c = c(t, x)) is Lipschitz continuous in (t, x).
- 7) The stage cost(L) and the terminal cost(g) are bounded ahead.

Assumption 1 guarantees the existence of a unique solution to the HJ PDEs, presented in Section III. Notably

$$\vartheta_1(t,x) = \inf_{\alpha \in \mathcal{A}(t)} \max_{\tau \in [t,T]} \int_t^\tau L(s, \mathbf{x}(s), \alpha(s)) \mathrm{d}s + g(\tau, \mathbf{x}(\tau))$$
(5)

subject to 
$$c(s, \mathbf{x}(s)) \le 0, \quad s \in [t, \tau]$$
 (6)

in which the state constraint  $c(s, \mathbf{x}(s)) \leq 0$  is satisfied in  $[t, \tau]$ instead of [t, T] as in (4) [1]. If the optimal value  $\vartheta_1(t, x)$  is finite, the state constraint has to be satisfied for all time in [t, T]. Otherwise, the time maximizer  $\tau$  will choose the time when the state constraint is not satisfied so that the optimal value becomes infinity.

State-constrained reach-avoid problem (SCRAP): For given initial time and state (t, x), solve

$$\vartheta_2(t,x) := \inf_{\alpha \in \mathcal{A}(t)} \min_{\tau \in [t,T]} \int_t^\tau L(s, \mathbf{x}(s), \alpha(s)) \mathrm{d}s + g(\tau, \mathbf{x}(\tau))$$

subject to 
$$c(s, \mathbf{x}(s)) < 0, \quad s \in [t, \tau]$$
 (8)

where x solves (1), and L, g, f, c are defined in the same way as for the SCCIP. Also, for each (t, x),  $\vartheta_2(t, x) \in \mathbb{R} \cup \{\infty\}$ , meaning that if there is no feasible control signal to satisfy the state constraint (8), then  $\vartheta_2(t, x)$  is infinite. As for SCRAP, we assume Assumption 1 holds.

## III. HJ EQUATIONS FOR STATE-CONSTRAINED REACHABILITY PROBLEMS

This section reviews the previously presented HJ PDEs for SCCIP and SCRAP [1], [2], which will be utilized to derive Hopf–Lax formulae in Section V. For each problem, this section

presents two HJ PDEs, one for the time-varying case and a second for the time-invariant case. The time-varying case features time-varying cost functions, dynamics, and state constraints.

#### A. HJ Equation for SCCIP

Lee and Tomlin [1] utilize the epigraphical technique to derive an HJ PDE whose solution characterizes the epigraph of the optimal cost ( $\vartheta_1$ ) for the SCCIP. In this formulation, we first encode the cost and constraint of the SCCIP into a state-augmented value function  $V_1$  (10) whose subzero-level set is the epigraph of the optimal cost ( $\vartheta_1$ ) for SCCIP

$$epi(\vartheta_1(t, \cdot)) := \{(x, z) \mid z \ge \vartheta_1(t, x)\}$$
$$= \{(x, z) \mid V_1(t, x, z) \le 0\}.$$
(9)

Then, the dynamic programming principle is applied to derive the HJ PDE for  $V_1$ .

Define the augmented value function  $V_1 = V_1(t, x, z)$ :  $[0, T] \times \mathbb{R}^n \times \mathbb{R} \to \mathbb{R}$ 

$$V_1(t, x, z) := \inf_{\alpha \in \mathcal{A}(t)} \max\left\{ \max_{s \in [t, T]} c(s, \mathbf{x}(s)) \right.$$
$$\left. \max_{\tau \in [t, T]} \int_t^\tau L(s, \mathbf{x}(s), \alpha(s)) \mathrm{d}s + g(\tau, \mathbf{x}(\tau)) - z \right\}$$
(10)

where x solves (1), (t, x) are initial time and states, and z is a new scalar variable that represents a value axis for the epigraph of  $\vartheta_1$ . Now, we will consider (x, z) as an augmented state in  $V_1$ .

 $V_1$  is continuous in (t, x, z)-space, and standard viscosity theory works for  $V_1$ . Theorem 1 presents HJ PDEs for  $V_1$  and finds  $\vartheta_1$  from  $V_1$ .

Theorem 1. (HJ PDE for SCCIP [1]): Suppose Assumption 1 holds.  $V_1$  in (10) is the unique viscosity solution to the HJ PDE

$$\max\left\{c(t,x) - V_1(t,x,z), g(t,x) - z - V_1(t,x,z)\right\}$$
$$\frac{\partial V_1}{\partial t} - \bar{H}\left(t,x,z,\frac{\partial V_1}{\partial x},\frac{\partial V_1}{\partial z}\right)\right\} = 0 \quad (11)$$

in  $(0,T) \times \mathbb{R}^n \times \mathbb{R}$ , where  $\overline{H} : [0,T] \times \mathbb{R}^n \times \mathbb{R} \times \mathbb{R}^n \times \mathbb{R} \to \mathbb{R}$  is given by

$$\bar{H}(t, x, z, p, q) := \max_{a \in A} [-p \cdot f(t, x, a) + qL(t, x, a)]$$
(12)

where p and q represent the gradients  $\frac{\partial V_1}{\partial x}$  and  $\frac{\partial V_1}{\partial z}$ , and

$$V_1(T, x, z) = \max\{c(T, x), g(T, x) - z\}$$
(13)

on  $\{t = T\} \times \mathbb{R}^n \times \mathbb{R}$ .

For the time-invariant case, the above HJ PDE (11) simplifies to

$$\max\left\{c(x) - V_1(t, x, z), \frac{\partial V_1}{\partial t} - \bar{H}_1^{\mathrm{TI}}\left(x, z, \frac{\partial V_1}{\partial x}, \frac{\partial V_1}{\partial z}\right)\right\} = 0$$
(14)

in  $(0,T) \times \mathbb{R}^n \times \mathbb{R}$ , where

$$\bar{H}_{1}^{\mathrm{TI}}(x, z, p, q) := \min\left\{0, \bar{H}(x, z, p, q)\right\}$$
(15)

for  $(x, z, p, q) \in \mathbb{R}^n \times \mathbb{R} \times \mathbb{R}^n \times \mathbb{R}$ , and  $\overline{H}$  is defined in (12).

Then

$$\vartheta_1(t,x) = \min z$$
 subject to  $V_1(t,x,z) \le 0.$  (16)

## B. HJ Equation for SCRAP

The HJ analysis for SCRAP is similar to that for SCCIP. We first define a state-augmented value function  $V_2$  (17) that combines the cost (7) and the constraint (8) of  $\vartheta_2$  so that  $V_2$ 's subzero-level set is the epigraph of the optimal cost  $\vartheta_2$  for the SCRAP. For  $(t, x, z) \in [0, T] \times \mathbb{R}^n \times \mathbb{R}$ 

$$V_{2}(t, x, z) := \inf_{\alpha \in \mathcal{A}(t)} \min_{\tau \in [t, T]} \max \left\{ \max_{s \in [t, \tau]} c(s, \mathbf{x}(s)) \right.$$
$$\int_{t}^{\tau} L(s, \mathbf{x}(s), \alpha(s)) \mathrm{d}s + g(\tau, \mathbf{x}(\tau)) - z \right\}$$
(17)

where x solves (1), (t, x) are initial time and states, and z is again the new variable that represents a value axis for the epigraph of  $\vartheta_2$ .

Theorem 2 presents HJ PDEs for  $V_2$  and finds  $\vartheta_2$  from  $V_2$ .

*Theorem 2.* (*HJ PDE for SCRAP* [2]): Suppose Assumption 1 holds.  $V_2$  in (17) is the unique viscosity solution to the HJ PDE

$$\max\left\{c(t,x) - V_2(t,x,z), \min\left\{g(t,x) - z - V_2(t,x,z), \frac{\partial V_2}{\partial t} - \bar{H}\left(t,x,z,\frac{\partial V_2}{\partial x},\frac{\partial V_2}{\partial z}\right)\right\}\right\} = 0$$
(18)

in  $(0,T) \times \mathbb{R}^n \times \mathbb{R}$ , where  $\overline{H}$  is defined in (12), and

$$V_2(T, x, z) = \max\{c(T, x), g(T, x) - z\}$$
(19)

on  $\{t = T\} \times \mathbb{R}^n \times \mathbb{R}$ .

For the time-invariant case,  $V_2$  is also the unique viscosity solution to HJ PDE

$$\max\left\{c(x) - V_2(t, x, z), \frac{\partial V_2}{\partial t} - \bar{H}_2^{\mathrm{TI}}(x, z, \frac{\partial V_2}{\partial x}, \frac{\partial V_2}{\partial z})\right\} = 0$$
(20)

in  $(0,T) \times \mathbb{R}^n \times \mathbb{R}$ , where

$$\bar{H}_{2}^{\mathrm{TI}}(x, z, p, q) = \max\left\{0, \bar{H}(x, z, p, q)\right\}$$
 (21)

for  $(x, z, p, q) \in \mathbb{R}^n \times \mathbb{R} \times \mathbb{R}^n \times \mathbb{R}$ ,  $\overline{H}$  is defined in (12), and

$$V_2(T, x, z) = \max\{c(x), g(x) - z\}$$
(22)

on  $\{t = T\} \times \mathbb{R}^n \times \mathbb{R}$ .

Then

$$\vartheta_2(t,x) = \min z$$
 subject to  $V_2(t,x,z) \le 0.$  (23)

Building on the HJ PDEs in this section, Section V presents the main contribution of this article: the proposal of Hopf–Lax formulae for SCCIP and SCRAP. Our Hopf–Lax formulae in Section V will assume convex Hamiltonians in the gradient space.  $\bar{H}$  (12) and  $\bar{H}_2^{\text{TI}}$  (20) are convex in (p, q), but  $\bar{H}_1^{\text{TI}}$ (14) is not. Thus, we do not have the Hopf–Lax formula for the time-invariant SCCIP. Section V presents three Hopf–Lax formulae for time-varying SCCIP, time-varying SCRAP, and time-invariant SCRAP.

## IV. VISCOSITY THEORY FOR TWO DIFFERENT PDES' SOLUTION EQUIVALENCE

This section proposes a general theorem in viscosity theory to investigate the equivalence of two first-order PDEs. Building on our new theory in this section, we will present and prove our main result, Hopf–Lax formulae for SCCIP and SCRAP, in Section V. We use a more general notation in this section.

Consider time  $t \in [0, T]$ , a state that consists of  $x \in \mathbb{R}^{n_x}$ ,  $z \in \mathbb{R}^{n_z}$ , and two value functions  $X_i = X_i(t, x, z) \in \mathbb{R}$  (i = 1, 2). Suppose  $X_i$  solves a first-order PDE

$$0 = F_i\left(t, x, z, X_i(t, x, z), \frac{\partial X_i}{\partial t}(t, x, z), \frac{\partial X_i}{\partial x}(t, x, z), \frac{\partial X_i}{\partial z}(t, x, z)\right)$$
(24)

in  $(0,T) \times \mathbb{R}^{n_x} \times \mathbb{R}^{n_z}$  for i = 1, 2, and the terminal values for  $X_1$  and  $X_2$  are the same as  $l = l(x, z) \in \mathbb{R}$ 

$$X_1(T, x, z) = X_2(T, x, z) = l(x, z) \quad \forall (x, z) \in \mathbb{R}^{n_x} \times \mathbb{R}^{n_z}.$$
(25)

We say that  $X_i = X_i(t, x, z)$  (i = 1, 2) is the viscosity solution of  $F_i$  [47, Ch. 10] if 1)  $X_i(T, x, z) = l(x, z)$  and, 2) for each smooth function  $U : (0, T) \times \mathbb{R}^{n_x} \times \mathbb{R}^{n_z} \to \mathbb{R}$ 

1) if  $X_i - U$  has a local maximum at a point  $(t_0, x_0, z_0) \in (0, T) \times \mathbb{R}^{n_x} \times \mathbb{R}^{n_z}$  and  $(X_i - U)(t_0, x_0, z_0) = 0$ 

$$F_i\left(t_0, x_0, z_0, U(t_0, x_0, z_0), \frac{\partial U}{\partial t}(t_0, x_0, z_0), \frac{\partial U}{\partial x}(t_0, x_0, z_0), \frac{\partial U}{\partial z}(t_0, x_0, z_0)\right) \ge 0$$
(26)

2) if  $X_i - U$  has a local minimum at a point  $(t_0, x_0, z_0) \in (0, T) \times \mathbb{R}^{n_x} \times \mathbb{R}^{n_z}$  and  $(X_i - U)(t_0, x_0, z_0) = 0$ 

$$F_i\left(t_0, x_0, z_0, U(t_0, x_0, z_0), \frac{\partial U}{\partial t}(t_0, x_0, z_0), \frac{\partial U}{\partial x}(t_0, x_0, z_0), \frac{\partial U}{\partial z}(t_0, x_0, z_0)\right) \le 0.$$
(27)

For SCCIP,  $F_1$  refers to the HJ PDE (11), t to the initial time, x to the initial state,  $n_x$  to n,  $n_z$  to 1, and  $X_1$  to  $V_1$ .  $F_2$  refers to the HJ PDE (41) that will be introduced later in Section V-A,  $X_2$  to  $W_1$  (39) also in Section V-A. Also, l(x, z) refers to the terminal condition for  $V_1$  and  $W_1$ : max{c(T, x), g(T, x) - z}. For SCRAP, we have a similar notation matching rule with SCCIP.

In the notion of viscosity theory, we present conditions under which the two different PDEs  $F_1$  and  $F_2$  have the same solution. Define superdifferentials and subdifferentials of  $X_i$  (i = 1, 2) with respect to z: for each  $(t, x, z) \in [0, T] \times \mathbb{R}^{n_x} \times \mathbb{R}^{n_z}$ ,  $q \in \partial_z^+ X_i(t, x, z)$  (i = 1, 2) is a superdifferential with respect

to z if  

$$\partial_z^+ X_i(t, x, z)$$

$$:= \left\{ q \mid \limsup_{\bar{z} \to 0} \frac{X_i(t, x, z + \bar{z}) - X_i(t, x, z) - q \cdot \bar{z}}{\|\bar{z}\|} \leq 0 \right\}$$
(28)

and  $q \in \partial_z^- X_i(t, x, z)$  (i = 1, 2) is a subdifferential with respect to z if

$$\partial_{z}^{-} X_{i}(t, x, z) := \left\{ q \mid \limsup_{\bar{z} \to 0} \\ \frac{X_{i}(t, x, z + \bar{z}) - X_{i}(t, x, z) - q \cdot \bar{z}}{\|\bar{z}\|} \ge 0 \right\}.$$
 (29)

Theorem 3 states that if two different PDEs ( $F_1$  and  $F_2$ ) are the same in the superdifferential or subdifferential domains in z, the two PDEs' solutions are the same.

Theorem 3: Suppose each of the two first-order PDEs in (24) with the terminal value (25) for i = 1, 2 has the unique solution  $(X_i)$ . If, for all  $(t, x, z, X, r, p) \in [0, T] \times \mathbb{R}^{n_x} \times \mathbb{R}^{n_z} \times \mathbb{R} \times \mathbb{R} \times \mathbb{R}^{n_x}, q \in \partial_z^+ X_1(t, x, z) \cup \partial_z^- X_1(t, x, z)$ 

$$F_1(t, x, z, X, r, p, q) = F_2(t, x, z, X, r, p, q)$$
(30)

then  $X_1 \equiv X_2$ . Here, r, p, q are the costates with respect to t, x, and z, respectively.

*Proof:* See Appendix A.

#### V. HOPF–LAX FORMULAE FOR THE STATE-CONSTRAINED REACHABILITY PROBLEMS

For SCCIP and SCRAP, the HJ PDEs in Theorems 1 and 2 can be numerically solved by grid-based methods, such as level-set [23] and fast marching methods [59]. These methods require spatial and temporal discretization, which leads to exponential computational complexity in the state's dimension. Thus, it is intractable to utilize these grid-based methods for high-dimensional systems [6]. This article provides more discussion about the computational complexity in Section VII-C.

This section presents our main results to alleviate this computational complexity, Hopf–Lax formulae for SCCIP and SCRAP: Section V-A for SCCIP, Section V-B for SCRAP, and Section V-C for time-invariant SCRAP.

## A. Hopf-Lax Formula for SCCIP

Define, for initial time and state  $(t, x) \in [0, T] \times \mathbb{R}^n$ 

$$\varphi_1(t,x) := \inf_{\beta} \max_{\tau \in [t,T]} \int_t^\tau H^*(s, \mathbf{x}(s), \beta(s)) \mathrm{d}s + g(\tau, \mathbf{x}(\tau))$$

subject to 
$$\begin{cases} \dot{\mathbf{x}}(s) = -\beta(s), s \in [t, T] \\ \beta(s) \in \operatorname{co}(\{-f(s, \mathbf{x}(s), a) \mid a \in A\}), s \in [t, T] \\ \mathbf{x}(t) = x \\ c(s, \mathbf{x}(s)) \le 0, s \in [t, T] \end{cases}$$
(32)

where  $\beta : [t,T] \to \mathbb{R}^n$  is a new measurable control signal,  $H : [0,T] \times \mathbb{R}^n \times \mathbb{R}^n \to \mathbb{R}$ 

$$H(s, x, p) := \bar{H}(s, x, z, p, -1)$$
  
=  $\max_{a \in A} [-p \cdot f(s, x, a) - L(s, x, a)]$  (33)

where p is the costate with respect to x, H is defined in (12), and  $H^*: [0,T] \times \mathbb{R}^n \times \mathbb{R}^n \to \mathbb{R} \cup \{\infty\}$  is the Legendre–Fenchel transformation of H with respect to p

$$H^*(s, x, b) := \max_{p \in \mathbb{R}^n} [p \cdot b - H(s, x, p)]$$
 (34)

where b is a new control input (b and p are dual variables). For each s and x,  $H^*(s, x, b)$  is finite if  $b \in co(\{-f(s, x(s), a) \mid a \in A\})$  but, otherwise, is infinite.

Theorem 4. (Hopf-Lax formula for SCCIP): For all  $(t, x) \in [0, T] \times \mathbb{R}^n$ 

$$\vartheta_1(t,x) = \varphi_1(t,x) \tag{35}$$

where  $\vartheta_1$  is the optimal cost of SCCIP [(3) subject to (4)], and  $\varphi_1$  is the Hopf–Lax formula for SCCIP [(31) subject to (32)].

We will utilize our proposed theory in Section IV to prove Theorem 4. We first investigate the superdifferentials and subdifferentials of  $V_1$  with respect to z.

Lemma 1. (Convexity of the value function in z): For each  $(t,x) \in [0,T] \times \mathbb{R}^n$ ,  $V_1$  in (10) is convex in  $z \in \mathbb{R}$ : for all  $z_1, z_2 \in \mathbb{R}$  and  $\theta_1, \theta_2 \in [0,1]$  such that  $\theta_1 + \theta_2 = 1$ 

$$V_1(t, x, \theta_1 z_1 + \theta_2 z_2) \le \theta_1 V_1(t, x, z_1) + \theta_2 V_1(t, x, z_2).$$
(36)

*Proof:* See Appendix **B**.

*Lemma 2:* For all  $(t, x, z) \in [0, T] \times \mathbb{R}^n \times \mathbb{R}$ 

$$\partial_z^- V_1(t, x, z) \subset [-1, 0] \tag{37}$$

and if  $\partial_z^+ V_1(t, x, z)$  is not the empty set, the set of superdifferentials with respect to z is a singleton

$$\partial_z^+ V_1(t, x, z) = \left\{ \frac{\partial V_1}{\partial z}(t, x, z) \right\} \subset [-1, 0].$$
(38)

Note that  $V_1$  is defined in (10).

*Proof:* See Appendix C.

(31)

By applying the HJ analysis in Section III-A, we have the following results for the Hopf–Lax formula for SCCIP [ $\varphi_1$  (31) subject to (32)]. Combining (31) and (32), define a value function in the augmented state space,  $W_1 : [0,T] \times \mathbb{R}^n \times \mathbb{R} \to \mathbb{R}$ :

$$W_1(t, x, z) = \inf_{\beta} \max\left\{ \max_{s \in [t, T]} c(s, \mathbf{x}(s)), \\ \max_{\tau \in [t, T]} \int_t^{\tau} H^*(s, \mathbf{x}(s), \beta(s)) \mathrm{d}s + g(\tau, \mathbf{x}(\tau)) - z \right\}$$
(39)

where  $\beta(s) \in \operatorname{co}(\{-f(s, \mathbf{x}(s), a) \mid a \in A\})$ . As shown in [51]

$$Dom(H^*(s, \mathbf{x}(s), \cdot)) = \{b \mid H^*(s, \mathbf{x}(s), b) < \infty\}$$
  
= co({-f(s, \mathbf{x}(s), a) | a \in A}). (40)

Thus, it is not necessary to add the control constraint in the infimum operation in (39) since  $H^*$  becomes infinity due to the control constraint (40).

By Theorem 1,  $W_1$  is the unique viscosity solution to

$$\max\left\{c(t,x) - W_1(t,x,z), g(t,x) - z - W_1(t,x,z), \frac{\partial W_1}{\partial t} - \bar{H}_W\left(t,x,z,\frac{\partial W_1}{\partial x},\frac{\partial W_1}{\partial z}\right)\right\} = 0 \quad (41)$$

in  $(0,T) \times \mathbb{R}^n \times \mathbb{R}$ , where  $\bar{H}_W : [0,T] \times \mathbb{R}^n \times \mathbb{R} \times \mathbb{R}^n \times \mathbb{R} \to \mathbb{R}$  is defined as

$$\bar{H}_W(t, x, z, p, q) := \max_b [p \cdot b + q H^*(t, x, b)]$$
 (42)

Here p and q represents the gradients  $\frac{\partial W_1}{\partial x}$  and  $\frac{\partial W_1}{\partial z}$ ,  $H^*$  is defined in (34), and

$$W_1(T, x, z) = \max\{c(T, x), g(T, x) - z\}$$
(43)

on  $\{t = T\} \times \mathbb{R}^n \times \mathbb{R}$ . Then

$$\varphi_1(t,x) = \min z$$
 subject to  $W_1(t,x,z) \le 0.$  (44)

Now, it is sufficient to prove  $V_1 \equiv W_1$ .

We will utilize Theorem 3 to prove  $V_1 \equiv W_1$ .  $F_1$  and  $F_2$  in Section IV refer to (11) and (41), respectively. Also, Lemma 3 analyzes the relationship between the two Hamiltonians:  $\bar{H}$  (12) for  $V_1$  and  $\bar{H}_W$  (42) for  $W_1$ .

*Lemma 3:* For 
$$(t, x, z, p, q) \in [0, T] \times \mathbb{R}^n \times \mathbb{R} \times \mathbb{R}^n \times \mathbb{R}$$

$$H(t, x, z, p, q) = H_W(t, x, z, p, q) \text{ if } q \le 0$$
 (45)

where  $\overline{H}$  and  $\overline{H}_W$  are defined in (12) and (42), respectively. *Proof:* See Appendix D.

Now, we are ready to conclude the proof of Theorem 4.

*Proof of Theorem 4:* Lemma 2 states that the subdifferentials and superdifferentials of  $V_1(t, x, z)$  with respect to z are less than or equal to 0 for all (t, x, z). Thus, by combining Theorem 3, Lemma 2, and Lemma 3, we prove  $V_1 \equiv W_1$ . By (16) and (44), we conclude  $\vartheta_1 \equiv \varphi_1$ .

# B. Hopf-Lax Formula for SCRAP

Define, for initial time and state  $(t, x) \in [0, T] \times \mathbb{R}^n$ 

$$\varphi_2(t,x) := \inf_{\beta} \min_{\tau \in [t,T]} \int_t^\tau H^*(s, \mathbf{x}(s), \beta(s)) \mathrm{d}s + g(\tau, \mathbf{x}(\tau))$$

subject to 
$$\begin{cases} \dot{\mathbf{x}}(s) = -\beta(s), s \in [t, T] \\ \beta(s) \in \operatorname{co}(\{-f(s, \mathbf{x}(s), a) \mid a \in A\}), s \in [t, T] \\ \mathbf{x}(t) = x \\ c(s, \mathbf{x}(s)) \le 0, s \in [t, \tau] \end{cases}$$
(47)

where the description for all variables is the same as in Section V-A.

Theorem 5. (Hopf–Lax formula for SCRAP): For all  $(t, x) \in [0, T] \times \mathbb{R}^n$ 

$$\vartheta_2(t,x) = \varphi_2(t,x) \tag{48}$$

where  $\vartheta_2$  is the optimal cost of SCRAP [(7) subject to (8)], and  $\varphi_2$  is the Hopf–Lax formula for SCRAP [(46) subject to (47)].

The proof of Theorem 5 similarly follows that of Theorem 4 except for the proof of the convexity of  $V_2$  in z-space, whose counterpart in the previous section is Lemma 1. Although  $V_2$  is convex in z-space, we have to prove it not in a similar way for Lemma 1. Our proof for the convexity of  $V_2$  is added to the proof of Lemma 1 in Appendix B (iii).

#### C. Hopf-Lax Formula for the Time-Invariant SCRAP

Define, for the initial time and state  $(t, x) \in [0, T] \times \mathbb{R}^n$ 

$$\varphi_{2}^{\text{TI}}(t,x) := \inf_{\beta} \int_{t}^{T} H_{2}^{\text{TI}^{*}}(\mathbf{x}(s),\beta(s))ds + g(\mathbf{x}(T))$$
(49)  
subject to 
$$\begin{cases} \dot{\mathbf{x}}(s) = -\beta(s), s \in [t,T] \\ \beta(s) \in \operatorname{co}(\{0\} \cup \{-f(\mathbf{x}(s),a) \mid a \in A\}), \\ s \in [t,T] \\ \mathbf{x}(t) = x \\ c(\mathbf{x}(s)) \leq 0, s \in [t,T] \end{cases}$$
(50)

where  $\beta : [t,T] \to \mathbb{R}^n$  is a new measurable control signal,  $H_2^{\text{TI}} : \mathbb{R}^n \times \mathbb{R}^n \to \mathbb{R}$ 

$$H_2^{\text{TI}}(x,p) := \bar{H}_2^{\text{TI}}(x,z,p,-1) = \max\{0, H(x,p)\}$$
$$= \max\{0, \max_{a \in A} [-p \cdot f(x,a) - L(x,a)]\}$$
(51)

where p is the costate with respect to x,  $\overline{H}_2^{\text{TI}}$  is defined in (21), H is defined in (33), and  $H_2^{\text{TI}*}: \mathbb{R}^n \times \mathbb{R}^n \to \mathbb{R} \cup \{\infty\}$  is the Legendre–Fenchel transformation of  $H_2^{\text{TI}}$  with respect to p

$$H_2^{\text{TI}*}(x,b) := \max_p [p \cdot b - H_2^{\text{TI}}(x,p)]$$
(52)

where b is a new control input.

(46)

Theorem 6. (Hopf-Lax formula for the time-invariant SCRAP): Consider the time-invariant SCRAP. For all  $(t, x) \in [0, T] \times \mathbb{R}^n$ 

$$\vartheta_2(t,x) = \varphi_2(t,x) = \varphi_2^{\mathrm{TI}}(t,x) \tag{53}$$

where  $\vartheta_2$  is the optimal cost of SCRAP [(7) subject to (8)],  $\varphi_2$  is the Hopf–Lax formula for SCRAP [(46) subject to (47)], and  $\varphi_2^{\text{TI}}$  is the Hopf–Lax formula for the time-invariant SCRAP [(49) subject to (50)].

We will combine Section IV and following formulations to prove Theorem 6. By applying the HJ analysis in Section III-B, we have the following results for the Hopf–Lax formula for the time-invariant SCCIP [ $\varphi_2$  (49) subject to (50)]. Combining (49) and (50), define a value function in the augmented state space.  $W_2^{\text{TI}} : [0, T] \times \mathbb{R}^n \times \mathbb{R} \to \mathbb{R}$ :

$$W_{2}^{\text{TI}}(t, x, z) = \inf_{\beta} \max\left\{ \max_{s \in [t, T]} c(\mathbf{x}(s)) \right. \\ \left. \int_{t}^{T} H_{2}^{\text{TI}*}(\mathbf{x}(s), \beta(s)) \mathrm{d}s + g(\mathbf{x}(T)) - z \right\}.$$
(54)

In (54), we can omit the control constraint in (50):  $\beta(s) \in$ co({0}  $\cup$  { $-f(\mathbf{x}(s), a) \mid a \in A$ }), since this control constraint is the domain of the control input  $\beta(s)$  for finite  $H_2^{\text{TI}*}(\mathbf{x}(s), \cdot)$ . *Lemma 4. (Domain of*  $H_2^{\text{TI}*}$ ): For all  $x \in \mathbb{R}^n$ 

$$Dom(H_2^{\text{TI}*}(x, \cdot)) = \{b \mid H_2^{\text{TI}*}(x, b) < \infty\}$$
$$= co(\{0\} \cup \{-f(x, a) \mid a \in A\})$$
(55)

where  $H_2^{\text{TI}*}$  is defined in (52).

*Proof:* See Appendix E.

By [18, Proposition 3.4],  $W_2^{\text{TI}}$  is the unique viscosity solution to

$$\max\left\{c(x) - W_2^{\mathrm{TI}}, \frac{\partial W_2^{\mathrm{TI}}}{\partial t} - \bar{H}_W^{\mathrm{TI}}\left(x, z, \frac{\partial W_2^{\mathrm{TI}}}{\partial x}, \frac{\partial W_2^{\mathrm{TI}}}{\partial z}\right)\right\} = 0 \quad (56)$$

in  $(0,T) \times \mathbb{R}^n \times \mathbb{R}$ , and  $W_2^{\text{TI}}(T,x,z) = \max\{c(x),g(x)-z\}$ on  $\{t=T\} \times \mathbb{R}^n \times \mathbb{R}$ , where  $\bar{H}_W^{\text{TI}} : \mathbb{R}^n \times \mathbb{R} \times \mathbb{R}^n \times \mathbb{R} \to \mathbb{R}$ 

$$\bar{H}_{W}^{\text{TI}}(x, z, p, q) := \max_{b} [p \cdot b + q H_{2}^{\text{TI*}}(x, b)]$$
(57)

where  $H_2^{\text{TI}*}$  is defined in (52), and p and q represents the gradients  $\frac{\partial W_2^{\Pi}}{\partial x}$  and  $\frac{\partial W_2^{\Pi}}{\partial z}$ . Then

$$\varphi_2^{\text{TI}}(t,x) = \min z \text{ subject to } W_2^{\text{TI}}(t,x,z) \le 0.$$
 (58)

Now, it is sufficient to show  $V_2 \equiv W_2^{\text{TI}}$ . To prove this, we will use Theorem 3 where we substitute (20) to  $F_1$  and (56) to  $F_2$ .

Lemma 5 analyzes the relationship between  $\bar{H}_2^{\text{TI}}$  (21) for  $V_2$  and  $\bar{H}_W^{\text{TI}}$  (57) for  $W_2^{\text{TI}}$ .

Lemma 5: For  $(\bar{x}, z, p, q) \in \mathbb{R}^n \times \mathbb{R} \times \mathbb{R}^n \times \mathbb{R}$  $\bar{H}_2^{\text{TI}}(x, z, p, q) = \bar{H}_W^{\text{TI}}(x, z, p, q) \text{ if } q \leq 0$  (59)

where  $\bar{H}_2^{\text{TI}}$  and  $\bar{H}_W^{\text{TI}}$  are defined in (21) and (57), respectively. *Proof:* See Appendix F.

Now, we are ready to conclude the proof of Theorem 6.

*Proof of Theorem 6:* In Theorem 3, we substitute the HJ PDE for  $V_2$  (20) to  $F_1$  and the HJ PDE for  $W_2^{\text{TI}}$  (56) to  $F_2$ . Since Lemmas 2 and 5 hold, Theorem 3 implies  $V_2 \equiv W_2^{\text{TI}}$ . By Theorem 5, (23), and (58), we conclude  $\vartheta_2 \equiv \varphi_2 \equiv \varphi_2^{\text{TI}}$ .

Although the time-varying case is more general than the time-invariant case, Section VI show that the Hopf–Lax formula for the time-invariant SCRAP has better convexity conditions. Thus, direct methods for the time-invariant SCRAP enable optimality guarantees and efficient computation for more classes of problems than for the time-varying SCRAP.

# VI. CONVEXITY ANALYSIS

We can efficiently solve SCCIP, SCRAP, and Hopf–Lax formulae by utilizing direct methods, such as multiple shooting [52] or collocation methods [53]. This section investigates convexity conditions under which direct methods guarantee optimality. The convexity analysis in this Section VI is valid regardless of any different numerical integration methods, including the midpoint, trapezoidal, and Simpson's rule, and numerical ODE solving methods, including linear multistep and Runge–Kutta methods [60]. For convenience, this section chooses a multiple shooting method that uses the first-order forward Euler method for both numerical integration and ODE solving.

We discretize the time interval to  $\{t_0 = 0, ..., t_K = T\}$ where  $\Delta_k := t_{k+1} - t_k$ . In the use of the first-order Euler method, the difference between the exact ODE solution and the Euler approximation becomes smaller for smaller  $\Delta_k$  [61], [62]. For notation, the state at  $t_k$  is x[k], and  $\alpha[k]$  and  $\beta[k]$  are control inputs at  $t_k$ . In this article, we use x to denote both a state trajectory in the continuous-time setting  $(x(\cdot))$  and a state sequence in the discrete-time setting  $(x[\cdot])$ . We apply the same notation rule for  $\alpha, \beta$ .

Section VI-A deals with SCCIP and its Hopf–Lax formula, and Section VI-B deals with SCRAP and its Hopf–Lax formulae for time-varying and time-invariant cases.

## A. Convexity Analysis for SCCIP and Its Hopf–Lax Formula

This subsection presents a convexity analysis for a temporally discretized SCCIP and Hopf–Lax formula. Consider the temporally discretized SCCIP

$$\vartheta_1(0,x) \simeq \min_{\mathbf{x}[\cdot],\alpha[\cdot]} \max_{k' \in \{0,\dots,K\}} \sum_{k=0}^{k'} L(t_k,\mathbf{x}[k],\alpha[k])\Delta_k + g(t_{k'},\mathbf{x}[k'])$$
(60)

subject to 
$$\begin{cases} \mathbf{x}[k+1] - \mathbf{x}[k] = \Delta_k f(t_k, \mathbf{x}[k], \alpha[k]) \\ k \in \{0, \dots, K-1\} \\ \alpha[k] \in A, \qquad k \in \{0, \dots, K-1\} \\ \mathbf{x}[0] = x \\ c(t_k, \mathbf{x}[k]) \le 0, \quad k \in \{0, \dots, K\}. \end{cases}$$
(61)

The temporal discretized SCCIP [(60) subject to (61)] is convex in  $(\mathbf{x}[\cdot], \alpha[\cdot])$ -space if Assumption 2 holds. Since a pointwise maximum of convex functions is convex, the cost in (60) is convex if  $\sum_{k=0}^{k'} L(t_k, \mathbf{x}[k], \alpha[k])\Delta_k + g(t_{k'}, \mathbf{x}[k'])$  is convex for each k'. Thus, the first two conditions in Assumption 2 are sufficient to have convex cost in (60). The other three conditions are for convex constraints (61).

Assumption 2. (Convexity conditions for the temporally discretized SCCIP).

1) L(t, x, a) is convex in (x, a) for all  $t \in [0, T]$ .

2) g(t, x) is convex in x for all  $t \in [0, T]$ .

- 3) c(t, x) is convex in x for all  $t \in [0, T]$ .
- 4) f(t, x, a) is affine in (x, a) for all  $t \in [0, T]$ .
- 5) A is convex.

We discretize the Hopf–Lax formula for SCCIP [ $\varphi_1$  in (31) subject to (32)] as the following:

$$\varphi_{1}(0,x) \simeq \min_{\mathbf{x}[\cdot],\beta[\cdot]} \max_{k' \in \{0,...,K\}} \sum_{k=0}^{k'} H^{*}(t_{k},\mathbf{x}[k],\beta[k])\Delta_{k} + g(t_{k'},\mathbf{x}[k'])$$
(62)

subject to 
$$\begin{cases} \mathbf{x}[k+1] - \mathbf{x}[k] = -\Delta_k \beta[k], k \in \{0, \dots, K-1\} \\ \beta[k] \in \mathbf{co}(\{-f(t_k, \mathbf{x}[k], a) \mid a \in A\}), \\ k \in \{0, \dots, K-1\} \\ \mathbf{x}[0] = x \\ c(t_k, \mathbf{x}[k]) \le 0, k \in \{0, \dots, K\}. \end{cases}$$

Based on the above convexity analysis for the temporally discretized SCCIP in  $(\mathbf{x}[\cdot], \beta[\cdot])$ -space, it is sufficient to verify convexity conditions for  $H^*(t, x, b)$  and  $\{(x, b) \mid b \in$  $\operatorname{co}(\{-f(t, x, a) \mid a \in A\})\}$  in (x, b). One of the authors' papers [51] provided sufficient conditions under which a finitedimensional optimization problem (62) subject to (63) is convex, as written ahead.

Lemma 6: Suppose 1)  $L(t, x, a) = L^x(t, x) + L^a(t, a)$ , and  $L^x$  is convex in x for each  $t \in [0, T]$ , 2)  $f(t, x, a) = M(t)x + L^a(t, a)$  for some matrix  $M(t) \in \mathbb{R}^n \times \mathbb{R}^n$ . Then,  $H^*(t, x, b)$  and  $\{(x, b) \mid b \in \operatorname{co}(\{-f(t, x, a) \mid a \in A\})\}$  are convex in (x, b) [51].

Combining Lemma 6 and the convexity analysis argument for SCCIP, we conclude sufficient convexity conditions for the temporally discretized Hopf–Lax formula for SCCIP in Assumption 3.

Assumption 3. (Convexity condition for the temporally discretized Hopf–Lax formula for SCCIP):

- 1)  $L(t, x, a) = L^{x}(t, x) + L^{a}(t, a)$  for some  $L^{x}$  and  $L^{a}$ , and  $L^{x}$  is convex in x for all  $t \in [0, T]$ .
- 2) g(t, x) is convex in x for all  $t \in [0, T]$ .
- 3) c(t, x) is convex in x for all  $t \in [0, T]$ .
- 4)  $f(t, x, a) = M(t)x + f^a(t, a)$  for some M and  $f^a$  for all  $t \in [0, T]$ .

Assumption 3 does not require 1) convex stage cost L in the control input a, 2) affine dynamics f in the control input a, and 3) convex control set A. This is because the Legendre–Fenchel transformation of the Hamiltonian  $(H^*)$  is always convex in the control input a, and the convex-hull operator (co) also convexifies the control-input space. We would like to note that the HJ analysis in Section III still needs to assume A is convex, as in Assumption 1.

*Remark 1:* The Hopf–Lax formula for SCCIP convexifies SCCIP in the control-input space.

## B. Convexity Analysis for SCRAP and Its Two Hopf–Lax Formulae

This section presents convexity analysis for SCRAP, the Hopf–Lax formula for the time-varying SCRAP, and the Hopf– Lax formula for the time-invariant SCRAP. The temporally discretized SCRAP is

$$\vartheta_2(0,x) \simeq \min_{\mathbf{x}[\cdot],\alpha[\cdot],k' \in \{0,\dots,K\}} \sum_{k=0}^{k'} L(t_k,\mathbf{x}[k],\alpha[k])\Delta_k + g(t_{k'},\mathbf{x}[k'])$$
(64)

subject to 
$$\begin{cases} \mathbf{x}[k+1] - \mathbf{x}[k] = \Delta_k f(t_k, \mathbf{x}[k], \alpha[k]), \\ k \in \{0, \dots, K-1\} \\ \alpha[k] \in A, k \in \{0, \dots, K-1\} \\ \mathbf{x}[0] = x \\ c(t_k, \mathbf{x}[k]) \le 0, k \in \{0, \dots, k'\}. \end{cases}$$
(65)

The cost in (64) is generally nonconvex in  $(\mathbf{x}[\cdot], \alpha[\cdot])$ -space since the pointwise minimum operator over k' is defined in a nonconvex set:  $\{0, \ldots, K\}$ . Thus, a sufficient convexity condition is that L and g are 0, as in Assumption 4. In this case, 0 is always a minimizer k'. Thus no additional conditions are necessary for convexity of the temporally discretized SCRAP.

Assumption 4. (Convexity condition for the temporally discretized SCRAP and its Hopf–Lax formula):

1) 
$$L \equiv g \equiv 0.$$

(63)

The temporally discretized Hopf–Lax formula for SCRAP [(46) subject to (47)] is as follows:

$$\varphi_{2}(0,x) \simeq \min_{\mathbf{x}[\cdot],\beta[\cdot],k' \in \{0,...,K\}} \sum_{k=0}^{k'} H^{*}(t_{k},\mathbf{x}[k],\beta[k])\Delta_{k} + g(t_{k'},\mathbf{x}[k'])$$
(66)

subject to 
$$\begin{cases} \mathbf{x}[k+1] - \mathbf{x}[k] = -\Delta_k \beta[k], k \in \{0, \dots, K-1\}\\ \beta[k] \in \mathbf{co}(\{-f(t_k, \mathbf{x}[k], a) \mid a \in A\}), \\ k \in \{0, \dots, K-1\}\\ \mathbf{x}[0] = x\\ c(t_k, \mathbf{x}[k]) \le 0, k \in \{0, \dots, k'\}. \end{cases}$$
(67)

By the same argument for SCRAP,  $H^*$  and g should be zero for the convexity of the discretized Hopf–Lax formula for SCRAP in  $(\mathbf{x}[\cdot], \beta[\cdot])$ -space. [51] proves that  $L \equiv 0$  implies  $H^* \equiv 0$ , thus, Assumption 4 is also a sufficient condition for convexity of the temporally discretized Hopf–Lax formula for SCRAP.

We discretize the Hopf–Lax formula for the time-invariant SCRAP [ $\varphi_2^{\text{TI}}$  in (49) subject to (50)]

$$\varphi_{2}^{\mathrm{TI}}(0,x) \simeq \min_{\mathbf{x}[\cdot],\beta[\cdot]} \sum_{k=0}^{K} H_{2}^{\mathrm{TI}*}(\mathbf{x}[k],\beta[k])\Delta_{k} + g(\mathbf{x}(K)) \quad (68)$$
subject to
$$\begin{cases} \mathbf{x}[k+1] - \mathbf{x}[k] = -\Delta_{k}\beta[k], \\ k \in \{0, \dots, K-1\} \\ \beta[k] \in \mathbf{co}(\{0\} \cup \{-f(\mathbf{x}[k],a) \mid a \in A\}), \\ k \in \{0, \dots, K-1\} \\ \mathbf{x}[0] = x \\ c(\mathbf{x}[k]) \le 0, k \in \{0, \dots, K\}. \end{cases}$$
(69)

We will prove that Assumption 5 is sufficient for the temporally discretized time-invariant SCRAP to be convex in  $(\mathbf{x}[\cdot], \beta[\cdot])$ -space. We will first show  $H_2^{\text{TI}*}(x, b)$  and  $\{(x, b) \mid b \in \text{co}(\{0\} \cup \{-f(x, a) \mid a \in A\})\}$  are convex in (x, b) if Assumption 5 holds. According to the definition of  $H_2^{\text{TI}}$  (51),  $H_2^{\text{TI}}$  does not have state dependency since f and L do not have it. Thus,  $H_2^{\text{TI}*}$  also does not have state dependency and is always convex in



Fig. 1. (a) SCCIP problem setting that describes the target positions  $x_g^r$  and constrained regions  $S_r$ , r = 1, ..., 4. (b)–(h) Optimal trajectories for the SCCIP problem at each time, where the blue circles are the vehicles' positions, and the black curves are optimal state trajectories. The SCCIP cost is maximized at 1.1 s. (a) Problem setting. (b) 0 s. (c) 1 s. (d) 1.1 s: Maximizer. (e) 2 s. (f) 3 s. (g) 4 s. (h) 5 s.

b. Similarly,  $\{(x, b) | b \in co(\{0\} \cup \{-f(x, a) | a \in A\})\}$  does not have state dependency and is convex in b. Finally, the second and fourth conditions in Assumption 5 imply the terminal cost and the state constraint are convex.

Assumption 5. (Convexity condition for the temporally discretized Hopf–Lax formula for the time-invariant SCRAP):

1) 
$$L(x, a) = L(a)$$
.

- 2) g(x) is convex in x.
- 3) c(x) is convex in x.

4) 
$$f(x, a) = f(a)$$
.

*Remark 2:* SCRAP and its Hopf–Lax formula for the timevarying case are generally nonconvex. However, the Hopf–Lax formula for the time-invariant SCRAP is convex if Assumption 5 holds. Assumption 5 does not require any convexity conditions in the control-input space.

#### **VII. NUMERICAL EXAMPLES**

This section provides two numerical examples to demonstrate the Hopf–Lax formulae for SCCIP and SCRAP. For numerical computation, we utilize the interior-point method [54, Ch. 11] in MATLAB, and a computer with an Apple M1 Pro and 16-GB RAM was used.

#### A. SCCIP: Robust Formation Control

Consider a 16-D system, which consists of four 4-D vehicles:  $\mathbf{x}^r(s) = (\mathbf{x}_1^r(s), \mathbf{x}_2^r(s), \mathbf{x}_3^r(s), \mathbf{x}_4^r(s)) \in \mathbb{R}^4$  refers to the *r*th vehicle's 4-D state  $(r = 1, \ldots, 4)$  at time *s*, and  $\mathbf{x}(s) = (\mathbf{x}^1(s), \ldots, \mathbf{x}^4(s)) \in \mathbb{R}^{16}$  refers to the system state at *s*. Each vehicle has two control inputs:  $\alpha^r(s) = (\alpha_1^r(s), \alpha_2^r(s)) \in \mathbb{R}^2$ . The system dynamics is

$$\dot{\mathbf{x}}_{1}^{r}(s) = \mathbf{x}_{2}^{r}(s), \quad \dot{\mathbf{x}}_{2}^{r}(s) = \alpha_{1}^{r}(s) \cos \alpha_{2}^{r}(s)$$

$$\dot{\mathbf{x}}_{3}^{r}(s) = \mathbf{x}_{4}^{r}(s), \quad \dot{\mathbf{x}}_{4}^{r}(s) = \alpha_{1}^{r}(s)\sin\alpha_{2}^{r}(s)$$
 (70)

where  $x_1^r$  and  $x_2^r$  ( $x_3^r$  and  $x_4^r$ ) are horizontal (vertical) position and velocity of the *r*th vehicle,  $\alpha_1^r$  is the magnitude of acceleration, and  $\alpha_2^r$  is the angle of the acceleration for r = 1, ..., 4.

As shown in Fig. 1(a), we would like to control the multivehicle system where the *r*th vehicle stays close to the point  $x_g^r$  within the interval  $s \in [0, 5]$  while maneuvering in the *r*th square-shaped constrained region  $(S_r)$ . We control the accelerations of the system (70). Due to initial velocities, it is challenging to control the *r*th vehicle to stay in  $S_r$  and close to  $x_g^r$ . For this problem, we solve

$$\inf_{\alpha} \max_{\tau \in [0,5]} \max_{r=1,\dots,4} \| (\mathbf{x}_{1}^{r}(\tau), \mathbf{x}_{3}^{r}(\tau)) - \mathbf{x}_{g}^{r} \|_{2}$$
(71)  
subject to 
$$\begin{cases} (70), \alpha^{r}(s) \in [-1,1] \times [-\frac{\pi}{6}, \frac{\pi}{6}] \\ \mathbf{x}(0) = \mathbf{x} \\ (\mathbf{x}_{1}^{r}(s), \mathbf{x}_{3}^{r}(s)) \in S_{r}, \quad r = 1, \dots, 4, s \in [0,5] \end{cases}$$

where the target positions are following:  $x_g^1 = (-1, 1), x_g^2 = (1, 1), x_g^3 = (1, -1), x_g^4 = (-1, -1)$ . The stage cost *L* is zero, and the terminal cost *g* is  $\max_{r=1,\dots,4} \| (\mathbf{x}_1^r(\tau), \mathbf{x}_3^r(\tau)) - x_g^r \|_2$ . For this SCCIP, the Hamiltonian *H* in (33) becomes

$$H(s, x, p) = \sum_{r=1}^{4} -p_1^r x_2^r - p_3^r x_4^r + \max\left\{ \|(p_2^r, p_4^r)\|_2, |p_2^r| \frac{\sqrt{3}}{2} + |p_4^r| \frac{1}{2} \right\}$$
(73)

where  $x_i^r$  is the name of the variable for which we substitute  $x_i^r(s)$  (r = 1, ..., 4),  $p_i^r$  is the costate with respect to  $x_i^r$ ,  $p^r = (p_1^r, p_2^r, p_3^r, p_4^r) \in \mathbb{R}^4$ , and  $p = (p^1, ..., p^4) \in \mathbb{R}^{16}$ . *H* is convex

(72)

in p, and any supporting hyperplane for H can be written as  $b \cdot p = 0$  for some normal vector  $b \in \mathbb{R}^{16}$ . Since the supporting hyperplane  $b \cdot p = 0$  passes through the origin for any b, the Legendre–Fenchel transformation of the Hamiltonian becomes

$$H^{*}(s, x, b) = \begin{cases} 0, & b_{1}^{r} = -x_{2}^{r}, b_{3}^{r} = -x_{4}^{r}, \|(b_{2}^{r}, b_{4}^{r})\|_{2} \le 1, |b_{r}^{4}| \le \frac{1}{2} \\ \infty, & \text{otherwise} \end{cases}$$
(74)

where  $b = (b_1^1, b_2^1, ..., b_4^4) \in \mathbb{R}^{16}$ , and  $\text{Dom}(H^*(s, x, \cdot))$  is analytically derived by (40). In general, if the stage cost L is zero,  $H^*$  becomes zero, which has been investigated in [51].

The Hopf–Lax formula for SCCIP in Theorem 1 is

$$\inf_{\beta} \max_{\tau \in [0,5]} \max_{r=1,\dots,4} \| (\mathbf{x}_{1}^{r}(\tau), \mathbf{x}_{3}^{r}(\tau)) - \mathbf{x}_{g}^{r} \|_{2}$$
(75)  
subject to  
$$\begin{cases} \dot{\mathbf{x}}^{r}(s) = -\beta^{r}(s) \\ \beta_{1}^{r}(s) = -\mathbf{x}_{2}^{r}(s), \beta_{3}^{r}(s) = -\mathbf{x}_{4}^{r}(s) \\ \| (\beta_{2}^{r}(s), \beta_{4}^{r}(s)) \|_{2} \le 1, |\beta_{4}^{r}(s)| \le \frac{1}{2} \\ \mathbf{x}(0) = \mathbf{x} \\ (\mathbf{x}_{1}^{r}(s), \mathbf{x}_{3}^{r}(s)) \in S_{r}, \quad r = 1, \dots, 4, s \in [0, 5] \end{cases}$$
(76)

where we get the second and third lines in (76) by substituting x(s) into x and  $\beta(s) = (\beta_1^1(s), \dots, \beta_4^4(s))$  into b in (74).

Consider the temporally discretized SCCIP and Hopf–Lax formula for SCCIP on any temporal discretization  $\{t_0 = 0, \dots, t_K = 5\}$  as in Section VI-A. The temporally discretized SCCIP is nonconvex, but the temporally discretized Hopf–Lax formula is convex since Assumption 3 is satisfied. Thus, direct methods with numerical optimization methods provide an optimal solution for the proposed Hopf–Lax formula for SCCIP.

For numerical computation of the Hopf–Lax formula, we discretize the temporal space to  $\{t_0 = 0, \ldots, t_{50} = 5\}$  with  $\Delta_k = 0.1$  (51 time steps). We utilize the interior-point method to solve the temporally discretized Hopf–Lax formula for SCCIP. The computation time is 108.5 s. This system is 16-D with four vehicles, for which it is intractable to utilize grid-based methods (such as the level-set method [23]) to solve the HJ PDEs (11).

As shown in Fig. 1(b)–(h), the four vehicles successfully stay in the constrained regions and stay near their target positions  $(x_{q}^{r})$ . Within the time interval, the maximum distance between  $x_q^r$  and the four vehicles is maximized at 1.1 s as shown in Fig. 1(d). At this time, each distance between  $x_q^r$  and the rth vehicle is 1.18, 0.43 0.42, and 1.16 m, and the first vehicle shows the farthest distance among the four vehicles. Thus, the optimal cost for SCCIP is 1.18. This means that all vehicles stay near their goals while maintaining less than 1.18-m distance within the time horizon while staying in their constraint regions  $S_r$  for all  $r = 1, \dots, 4$ . The first vehicle's trajectory shown in Fig. 1(b)–(h) initially moves downward, with a velocity in the -y-direction. Thus, in the solution found by SCCIP, the optimal control first decreases its vertical speed to ensure that it remains in  $S_1$ . Then, it heads toward  $x_a^1$ , in order to stay close to  $x_a^1$ , as incentivized by the cost function.

This example shows an optimal-control analysis for SCCIP, and this can be additionally utilized in decision-making for hardware specification. For example, consider designing multiple mobile manipulators for which we attach a manipulator to each mobile robot (vehicle). Each manipulator aims to perform some tasks at its target position  $x_g^r$  for all time in [0,5]. Our SCCIP analysis provides a guideline for choosing the manipulator's workspace that has to cover more than 1.18 m since all vehicles can be controlled to stay within 1.18-m-radius regions from  $x_g^r$ for all time in [0,5].

#### B. Hopf-Lax Formula for the Time-Invariant SCRAP

Consider a 12-D system, which consists of six 2-D vehicles:  $\mathbf{x}^{r}(s) = (\mathbf{x}_{1}^{r}(s), \mathbf{x}_{2}^{r}(s)) \in \mathbb{R}^{2}$  refers to the *r*th vehicle's 2-D state (r = 1, ..., 6) at time *s*, and  $\mathbf{x}(s) = (\mathbf{x}^{1}(s), ..., \mathbf{x}^{6}(s)) \in \mathbb{R}^{12}$ . Each vehicle has two control inputs:  $\alpha^{r}(s) = (\alpha_{1}^{r}(s), \alpha_{2}^{r}(s)) \in \mathbb{R}^{2}$ . The system dynamics is

$$\dot{\mathbf{x}}_1^r(s) = \alpha_1^r(s) + 2, \quad \dot{\mathbf{x}}_2^r(s) = \alpha_2^r(s) + 1$$
 (77)

where  $x_1^r$  and  $x_2^r$  are horizontal and vertical positions of the *r*th vehicle at *s*, and  $\alpha_1^r(s)$  and  $\alpha_2^r(s)$  are horizontal and vertical velocities, respectively.

As shown in Fig. 2(a), we would like to control the multivehicle system so that the distance between the 2-D center position of the six vehicles and the goal point is minimized while maintaining the hexagonal formation with 0.1-error bound

$$\begin{split} \inf_{\alpha} \min_{\tau \in [0,5]} \left\| \frac{\mathbf{x}^{1}(\tau) + \dots + \mathbf{x}^{6}(\tau)}{6} - x_{g} \right\|_{2} \tag{78} \\ \text{subject to} \begin{cases} (77), \|\alpha^{r}(s)\|_{\infty} \leq 0.5, r = 1, \dots, 6, s \in [0,5] \\ \mathbf{x}(0) = x \\ \|\mathbf{x}^{r}(s) - \frac{\mathbf{x}^{1}(\tau) + \dots + \mathbf{x}^{6}(\tau)}{6} - d^{r} \|_{\infty} \leq 0.1, s \in [0,\tau] \end{cases} \end{aligned}$$

where  $x_g = (1,0)$ , and  $d^r = 0.5(\cos((r-1)\pi/3), \sin((r-1)\pi/3))$  is a vector for the *r*th vehicle that indicates the formation direction from the 2-D center position of the six vehicles (r = 1, ..., 6).

For this problem, we will derive the Hopf–Lax formula for the time-invariant SCRAP since the temporally discretized Hopf–Lax formula for the time-varying SCRAP as in Section VI-B is generally nonconvex.

The Hamiltonian  $H_2^{\text{TI}}(x,p)$  (51) is

$$H_2^{\mathrm{TI}}(x,p) = \max\left\{0, \sum_{r=1}^{6} [-2p_1^r - p_2^r + \|p_1^r\| + \|p_2^r\|]\right\}$$
(80)

where  $x = (x_1^1, x_2^1, \dots, x_2^6) \in \mathbb{R}^{12}$ ,  $p_i^r$  is the costate with respect to  $x_i^r$ ,  $p^r = (p_1^r, p_2^r) \in \mathbb{R}^2$ , and  $p = (p^1, \dots, p^6) \in \mathbb{R}^{12}$ . Since  $H_2^{\text{TI}}$  is a pointwise maximum of two convex functions in p,  $H_2^{\text{TI}}$ is convex in p. Also, for all  $b \in \mathbb{R}^{12}$ , the supporting hyperplane of  $H_2^{\text{TI}}$  in p-space with respect to the normal vector b crosses the



Fig. 2. (a) SCRAP problem setting that describes the target position  $x_g$  and the six vehicles' 2-D positions. (b)–(h) Optimal trajectories for the SCRAP problem at each time, where the blue circles are the vehicles' positions, and the black curves are optimal state trajectories. The SCRAP cost is minimized at 2.3 s. (a) Problem setting. (b) 0 s. (c) 1 s. (d) 2 s. (e) 2.3 s: Minimizer. (f) 3 s. (g) 4 s. (h) 5 s.

origin. Thus

$$H_2^{\mathrm{TI}*}(x,b) = \begin{cases} 0, & \text{in } \mathrm{Dom}(H_2^{\mathrm{TI}*}(x,\cdot)) \\ \infty, & \text{otherwise} \end{cases}$$
(81)

where  $b = (b^1, \ldots, b^6) = (b_1^1, b_2^1, \ldots, b_1^6, b_2^6) \in \mathbb{R}^{12}$ . By Lemma 4,  $\text{Dom}(H_2^{\text{TI*}}(x, \cdot)) = \text{co}(\{0\} \cup \{-f(x, a) \mid a \in [-0.5, 0.5]^{12}\}) \subset \mathbb{R}^{12}$  is analytically derived as follows:

$$Dom(H_2^{\Pi*}(x,\cdot)) = \{b \mid \forall r_1, r_2 \in \{1, \dots, 6\},\$$
  
$$-2.5 \le b_1^{r_1}, -1.5 \le b_2^{r_1}, \frac{3}{5}b_1^{r_1} - b_1^{r_2} \ge 0, b_1^{r_1} - \frac{3}{5}b_1^{r_2} \le 0,\$$
  
$$b_1^{r_1} \ge 5b_2^{r_2}, b_1^{r_1} \le b_2^{r_2}, b_2^{r_2} \ge 3b_2^{r_1}, 3b_2^{r_2} \le b_2^{r_1}\}.$$
 (82)

The Hopf–Lax formula for the time-invariant SCRAP in Theorem 6 is

$$\inf_{\beta} \left\| \frac{\mathbf{x}^{1}(5) + \dots + \mathbf{x}^{6}(5)}{6} - x_{g} \right\|_{2}$$
(83)

subject to 
$$\begin{cases} \dot{\mathbf{x}}^{r}(s) = -\beta^{r}(s) \\ \beta(s) \in \text{Dom}(H_{2}^{\text{TI*}}(x, \cdot)) \text{ in (82)} \\ \mathbf{x}(0) = x \\ \|\mathbf{x}^{r}(s) - \frac{\mathbf{x}^{1}(s) + \dots + \mathbf{x}^{6}(s)}{6} - d^{r}\|_{\infty} \leq 0.1, s \in [0, 5]. \end{cases}$$
(84)

Consider the temporally discretized SCRAP and Hopf–Lax formula on any temporal discretization  $\{t_0 = 0, \ldots, t_K = 5\}$  as in Section VI-B. The temporally discretized SCRAP is nonconvex, but the temporally discretized Hopf–Lax formula for the time-invariant case is convex since Assumption 5 is satisfied, which enables direct methods to compute an optimal solution.

We utilized a numerical algorithm presented in [58] to find optimal control signals and state trajectories for the SCRAP problem. The computation time is 158.5 s. Fig. 2(b)–(h) shows the optimal state (position) trajectories of the multivehicle system. The distance between the center of the six vehicles and the goal position  $x_g$  is minimized at 2.3 s, as shown in Fig. 2(e), and then, the center of the six vehicles moves away from the goal position.

This example demonstrates our method's potential usefulness in multiple vehicle operations, for example, the control a spacecraft system to get close to an object of interest, such as an asteroid, and then the release of smaller exploration robots



Fig. 3. This figure shows the computation time of the level-set method and the time-invariant Hopf–Lax formula with different numbers of vehicles. Each vehicle is 2-D.

from the spacecraft to do more detailed sensing of the asteroid. The SCRAP analysis provides the optimal control and the time at which the exploration robots have to be released from the spacecraft.

## C. Computation Time

In this section, we compare the computation time of the Hopf– Lax formula for the time-invariant SCRAP and the level-set method. In the example in Section VII-B, we can easily change the number of vehicles. Fig. 3 shows the computation time where the number of vehicles is 1–10. We used the interior-point method [54, Ch. 11] to solve the proposed Hopf–Lax formula, which has polynomial–computational complexity in the state dimension, which is also demonstrated in Fig. 3. Since each vehicle is 2-D, the state dimension varies from 2 to 20.

The level-set method requires spatial and temporal discretization, which leads to exponential computational complexity in the state's dimension. The level-set method handles this problem if the number of vehicles is smaller than three. For numerical computation, we discretize the temporal space with 51 points and each axis of the augmented state (x, z) space with 61 points. The computation time is 26.9 s for the one-vehicle and 27.4 h  $(9.86 \times 10^4 \text{ s})$  for the two-vehicle.

The level-set method provides a closed-loop control. Thus, offline computation is allowed because a closed-loop control is robust to disturbances, measurement noise, and system modeling errors in practice. On the other hand, our Hopf–Lax methods provide an open-loop control, requiring real-time computation to be robust to the above factors.

Offline computation of the level-set method is intractable for high-dimensional systems due to the exponential growth of computational complexity in the state dimension and computing machines' memory limits. However, our Hopf–Lax method computes a solution even if the computation time might not meet real-time computation. For the three-vehicle setting (6-D), the level set method requires 641.1 TB of memories for a numerical solution to the HJ PDE, where 51 temporal discretization points and 61 spatial discretization points on each axis of (x, z) are used. It still requires 367.4 gigabytes if the number of spatial discretization points on each axis is reduced by 21. On the other hand, our Hopf–Lax method computes a solution even for a 10vehicle setting (20-D). In order to reduce the computation time, the Hopf–Lax formula can be incorporated with approximation methods, such as the receding-horizon technique [63], [64].

#### VIII. CONCLUSION

This article considers two state-constrained reachability problems: 1) the control invariance problem (SCCIP) and 2) reachavoid problem (SCRAP). We propose three Hopf-Lax formulae for 1) SCCIP, 2) SCRAP, and 3) the time-invariant SCRAP and provide proofs. In the proof of each Hopf-Lax formula, this article considers two HJ PDEs: One for SCCIP (or SCRAP) and the other for the corresponding Hopf-Lax formula and then proves these two different HJ PDEs have the same solution. This proof technique can be generally applied to other state-constrained optimal control problems' Hopf-Lax theory. This article also shows that Hopf-Lax formulae always have more favorable convexity conditions than SCCIP and SCRAP. Thus, direct methods efficiently compute solutions without losing optimality guarantees for more classes of problems. This benefit has been demonstrated via simulations compared with HJ analysis. The level-set method to solve HJ PDEs requires exponentially increasing memory space and computation time in the state dimension, so it is intractable to deal with more than 5-D systems. On the other hand, our method deals with 20-D and potentially much higher-dimensional systems without losing the optimality guarantee under the specified conditions.

## **APPENDIX**

## A. Proof of Theorem 3

i) The terminal values of  $X_1$  and  $X_2$  are the same.

ii) Consider a smooth function  $U : [0,T] \times \mathbb{R}^n \times \mathbb{R} \to \mathbb{R}$ . If  $X_1 - U$  has a local maximum at  $(t_0, x_0, z_0) \in (0,T) \times \mathbb{R}^n \times \mathbb{R}$  and  $(X_1 - U)(t_0, x_0, z_0) = 0$ 

$$F_1\left(t_0, x_0, z_0, U(t_0, x_0, z_0), \frac{\partial U}{\partial t}(t_0, x_0, z_0), \frac{\partial U}{\partial x}(t_0, x_0, z_0), \frac{\partial U}{\partial z}(t_0, x_0, z_0)\right) \ge 0.$$
(85)

Since  $X_1 - U$  has a local maximum at  $(t_0, x_0, z_0)$ ,  $\frac{\partial U}{\partial z}(t_0, x_0, z_0)$  is in  $\partial_z^+ X_1(t_0, x_0, z_0)$ . Then, by (30)

$$F_2\left(t_0, x_0, z_0, U(t_0, x_0, z_0), \frac{\partial U}{\partial t}(t_0, x_0, z_0), \frac{\partial U}{\partial x}(t_0, x_0, z_0), \frac{\partial U}{\partial z}(t_0, x_0, z_0)\right) \ge 0.$$
(86)

iii) Consider a smooth function  $U : [0,T] \times \mathbb{R}^n \times \mathbb{R} \to \mathbb{R}$ . If  $X_1 - U$  has a local minimum at  $(t_0, x_0, z_0) \in (0,T) \times \mathbb{R}^n \times \mathbb{R}$  and  $(X_1 - U)(t_0, x_0, z_0) = 0$ 

$$F_1\left(t_0, x_0, z_0, U(t_0, x_0, z_0), \frac{\partial U}{\partial t}(t_0, x_0, z_0), \frac{\partial U}{\partial x}(t_0, x_0, z_0), \frac{\partial U}{\partial z}(t_0, x_0, z_0)\right) \le 0.$$
(87)

Since  $X_1 - U$  has a local minimum at  $(t_0, x_0, z_0)$ ,  $\frac{\partial U}{\partial z}(t_0, x_0, z_0)$  is in  $\partial_z^- X_1(t_0, x_0, z_0)$ . By (30)

$$F_2\left(t_0, x_0, z_0, U(t_0, x_0, z_0), \frac{\partial U}{\partial t}(t_0, x_0, z_0), \frac{\partial U}{\partial x}(t_0, x_0, z_0), \frac{\partial U}{\partial z}(t_0, x_0, z_0)\right) \le 0.$$
(88)

#### B. Proof of Lemma 1

i) For all  $y_1, y_2, y_3, y_4 \in \mathbb{R}$ ,  $\max\{y_1 + y_2, y_3 + y_4\} \le \max\{y_1, y_3\} + \max\{y_2, y_4\}.$ (89)

ii) Proof of (36) for  $V_1$ . Let

$$J_c(\alpha, \tau) := \max_{s \in [t,T]} c(s, \mathbf{x}(s))$$
(90)

$$J_r(\alpha,\tau) := \int_t^\tau L(s,\mathbf{x}(s),\alpha(s)) \mathrm{d}s + g(\tau,\mathbf{x}(\tau)).$$
(91)

$$V_{1}(t, x, \theta_{1}z_{1} + \theta_{2}z_{2})$$

$$= \inf_{\alpha \in \mathcal{A}} \max_{\tau \in [t,T]} \max \left\{ \theta_{1}J_{c}(\alpha, \tau) + \theta_{2}J_{c}(\alpha, \tau) \right\}$$

$$= \inf_{\alpha \in \mathcal{A}} \max_{\tau \in [t,T]} \theta_{1} \max \left\{ J_{c}(\alpha, \tau), J_{r}(\alpha, \tau) - z_{1} \right\}$$

$$= \inf_{\alpha \in \mathcal{A}} \max_{\tau \in [t,T]} \left\{ J_{c}(\alpha, \tau), J_{r}(\alpha, \tau) - z_{2} \right\}.$$
(92)

The second inequality is by (89). For  $\alpha \in A$ , we use  $\tau_*(\alpha)$  to denote a maximizer of the last term in (92). By the triangular inequality, we simplify (92) to

$$V_{1}(t, x, \theta_{1}z_{1} + \theta_{2}z_{2})$$

$$\leq \inf_{\alpha \in \mathcal{A}} \theta_{1} \max \left\{ J_{c}(\alpha, \tau), J_{r}(\alpha, \tau_{*}(\alpha)) - z_{1} \right\}$$

$$+ \inf_{\alpha \in \mathcal{A}} \theta_{2} \max \left\{ J_{c}(\alpha, \tau), J_{r}(\alpha, \tau_{*}(\alpha)) - z_{2} \right\}$$

$$\leq \theta_{1}V_{1}(t, x, z_{1}) + \theta_{2}V_{1}(t, x, z_{2}). \tag{93}$$

The last inequality holds by the definition of  $V_1$  in (10).

iii) We additionally prove that  $V_2$  (17) is convex in z-space: for each  $(t, x) \in [0, T] \times \mathbb{R}^n$ , for all  $z_1, z_2 \in \mathbb{R}$  and  $\theta \in [0, 1]$ 

$$V_2(t, x, \theta_1 z_1 + \theta_2 z_2) \le \theta_1 V_2(t, x, z_1) + \theta_2 V_2(t, x, z_2).$$

Similar to (92)

$$V_{2}(t, x, \theta_{1}z_{1} + \theta_{2}z_{2})$$

$$\leq \inf_{\alpha \in \mathcal{A}} \min_{\tau \in [t,T]} \theta_{1} \max \left\{ J_{c}(\alpha, \tau), J_{r}(\alpha, \tau) - z_{1} \right\}$$

$$+ \theta_{2} \max \left\{ J_{c}(\alpha, \tau), J_{r}(\alpha, \tau) - z_{2} \right\}.$$
(94)

Since the last term in (94) is greater than or equal to  $\theta_1 V_2(t, x, z_1) + \theta_2 V_2(t, x, z_2)$  by the triangular inequality, we conclude the proof.

#### C. Proof of Lemma 2

i) The proof of (37).

For  $\overline{z} \ge 0$ ,  $V_1(t, x, z + \overline{z}) \le V_1(t, x, z)$ , and by the distributive property of the maximum operations

$$V_1(t, x, z + \bar{z}) = \inf_{\alpha \in \mathcal{A}} \max_{\tau \in [t, T]} \max \left\{ \max_{s \in [t, T]} c(s, \mathbf{x}(s)) + \bar{z}, \\ \int_t^\tau L(s, \mathbf{x}(s), u(s)) \mathrm{d}s + g(\tau, \mathbf{x}(\tau)) - z \right\} - \bar{z}$$
  
$$\geq V_1(t, x, z) - \bar{z}.$$

Thus, for  $\bar{z} \ge 0$ 

$$V_1(t, x, z) - \bar{z} \le V_1(t, x, z + \bar{z}) \le V_1(t, x, z)$$
(95)

and, by the same derivation, for  $\bar{z} \leq 0$ 

$$V_1(t, x, z) \le V_1(t, x, z + \bar{z}) \le V_1(t, x, z) - \bar{z}.$$
 (96)

Suppose there exists q > 0 in  $\partial_z^- V_1(t, x, z)$ . Then, there exists  $\epsilon > 0$  such that  $V_1(t, x, z + \overline{z}) \ge V_1(t, x, z) + q\overline{z}$  for all  $\overline{z} \in [-\epsilon, \epsilon]$ . However, for  $\overline{z} \in (0, \epsilon)$ 

$$V_1(t, x, z + \bar{z}) > V_1(t, x, z).$$
 (97)

This contradicts (95).

Suppose there exists q < -1 in  $\partial_z^- V_1(t, x, z)$ . Then, there exists  $\epsilon > 0$  such that  $V_1(t, x, z + \overline{z}) \ge V_1(t, x, z) + q\overline{z}$  for all  $\overline{z} \in [-\epsilon, \epsilon]$ . However, for  $\overline{z} \in (-\epsilon, 0)$ 

$$V_1(t, x, z + \bar{z}) > V_1(t, x, z) - \bar{z}.$$
 (98)

This contradicts (96). Thus,  $q \in [-1, 0]$ .

ii) The proof of (38).

The convexity of  $V_1(t, x, z)$  stated in Lemma 1 implies that  $\partial_z^+ V_1(t, x, z)$  contains a single superdifferential q if  $V_1(t, x, z)$  is locally affine in z, otherwise,  $\partial_z^+ V_1(t, x, z)$  is the empty set. If  $V_1(t, x, z)$  is locally affine in z, it is also differentiable in z. As  $\bar{z}$  converges to 0 in (95), we have  $\partial_z^+ V_1(t, x, z) = \{\frac{\partial V_1}{\partial z}(t, x, z)\} \subset [-1, 0].$ 

iii) We would like to note that the analogous derivation concludes that this lemma also holds for  $V_2: \partial_z^- V_2(t, x, z) \subset [-1, 0]$  for all  $(t, x, z) \in [0, T] \times \mathbb{R}^n \times \mathbb{R}$ , and if  $\partial_z^+ V_2(t, x, z)$  is not the empty set, then  $\partial_z^+ V_2(t, x, z) = \{\frac{\partial V_2}{\partial z}(t, x, z)\} \subset [-1, 0]$ .

#### D. Proof of Lemma 3

i) Case 1: q = 0.

ii) Case 2: a < 0

$$\bar{H}(t, x, z, p, 0) = \max_{a \in A} [-p \cdot f(t, x, a)]$$
  
= max{ $p \cdot b \mid b = -f(t, x, a), a \in A$ }. (99)

Since  $\{-f(t, x, a) \mid a \in A\} \subset co(\{-f(t, x, a) \mid a \in A\})$ 

$$H(t, x, z, p, 0) \le H_W(t, x, z, p, 0).$$
 (100)

On the other hand, let  $b_* \in \arg \max_{b \in \operatorname{co}(\{-f(t,x,a) \mid a \in A\})} [p \cdot b]$ . Since  $\{-f(t,x,a) \mid a \in A\}$  is compact, there exists a finite number of  $b_i \in \{-f(t,x,a) \mid a \in A\}$  and  $\theta_i \in [0,1]$  such that  $b_* = \sum_i \theta_i b_i$  and  $\sum_i \theta_i = 1$ . Then

$$\bar{H}_{W}(t, x, z, p, 0) = \sum_{i} \theta_{i} p \cdot b_{i} \leq \max_{i} \{ p \cdot b_{i} \}$$
$$\leq \max_{b \in \{-f(t, x, a) \mid a \in A\}} [p \cdot b] = \bar{H}(t, x, z, p, 0).$$
(101)

The last inequality holds since all  $b_i$ s are in  $\{-f(t, x, a) \mid a \in A\}$ . By (100) and (101), we have

$$\bar{H}(t,x,z,p,0) = \bar{H}_W(t,x,z,p,0).$$
 (102)

$$\bar{H}(t, x, z, p, q) = \max_{a \in A} -p \cdot f(t, x, a) + qL(t, x, a)$$

$$= -qH\left(t, x, -\frac{p}{q}\right).$$
(103)

Since *H* is convex in *p* for each (t, x) and lower semi-continuous in *p*,  $H^{**} \equiv H$ . Thus, we have

$$\bar{H}_{W}(t, x, z, p, q) = -q \max_{b \in \operatorname{co}(\{-f(t, x, a) \mid a \in A\})} \left[-\frac{p}{q} \cdot b - H^{*}(t, x, b)\right] \\
= -q H^{**}\left(t, x, -\frac{p}{q}\right) = -q H\left(t, x, -\frac{p}{q}\right). \quad (104)$$

By (103) and (104), we conclude

$$\bar{H}(t,x,z,p,q) = \bar{H}_W(t,x,z,p,q)$$
(105)

for all q < 0.

#### E. Proof of Lemma 4

This proof generalizes the proof of Lemma 1 [58], which is for the zero stage cost problem.

i) For 
$$b \in B(x)$$
,  $b = -f(x, \bar{a})$  for some  $\bar{a} \in A$ . Then  
 $H_2^{\text{TI}*}(x, b) = \max_p -p \cdot f(x, \bar{a}) - H_2^{\text{TI}}(x, p)$   
 $\leq \max_p -p \cdot f(x, \bar{a}) + \min_{a \in A} p \cdot f(x, a) + L(x, a)$   
 $< \infty.$  (106)

The last inequality holds since  $\min_{a \in A} p \cdot f(x, a) + L(x, a) \le p \cdot f(x, \bar{a}) + L(x, \bar{a})$  and L is finite for a fixed x. ii) If b = 0

$$H_2^{\text{TI}*}(x,b) = \max_p - H_2^{\text{TI}}(x,p) \le 0 < \infty.$$
(107)

There exists a finite set of  $\theta_i \in [0, 1]$   $(\sum_i \theta_i \le 1), a_i \in A$  such that  $b = -\sum_i \theta_i f(x, a_i)$ . Since  $H_2^{\text{TI}*}$  is convex in b

$$H_2^{\text{TI*}}(x,b) \le \sum_i \theta_i H_2^{\text{TI*}}(x,b_i) + (1 - \sum_i \theta_i) H_2^{\text{TI*}}(x,0) < \infty$$

by (106) and (107).

iv)  $b \notin \operatorname{co}(\{0\} \cup B(x))$ .

For the two convex sets  $\{b\}$  and  $co(\{0\} \cup B(x))$ , by the separating hyperplane theorem [54], there exists a hyperplane  $(P : \mathbb{R}^n \to \mathbb{R}): P(b') := p' \cdot b' + c$  such that P(b) > 0 but P(b') < 0 for all  $b' \in co(\{0\} \cup B(x))$ . By picking p = dp'

$$\begin{aligned} H_2^{\mathrm{TI}*}(x,b) &\geq \sup_d \min \bigg\{ dp' \cdot b, \min_{a \in A} dp' \cdot (b+f(x,a)) \\ &+ L(x,a) \bigg\}. \end{aligned}$$

Since  $p' \cdot b > 0$  and  $p' \cdot (b + f(x, a)) > 0$  for all  $a \in A$ , the supremum of the right term in the above equation is attained at  $d = \infty$ , thus,  $H_2^{\text{TI}*}(x, b) = \infty$ .

## F. Proof of Lemma 5

i) Case 1: 
$$q = 0$$
  
 $\bar{H}_2^{\text{TI}}(x, z, p, 0) = \max\{0, \max_{b \in B(x)} p \cdot b\}$  (108)

where B(x) is defined in (55). Since  $B(x) \subset co(\{0\} \cup B(x))$ 

$$\bar{H}_2^{\text{TI}}(x, z, p, 0) \le \bar{H}_W^{\text{TI}}(x, z, p, 0).$$
 (109)

On the other hand, let  $b_* \in \arg \max_{b \in \operatorname{co}(\{0\} \cup B(x))} p \cdot b$ , then, there exists a finite number of  $b_i \in B(x)$  and  $\theta_i \in [0, 1]$  such that  $b_* = \sum_i \theta_i b_i$  and  $\sum_i \theta_i < 1$ . Thus, we have

$$\bar{H}_W^{\mathrm{TI}}(x, z, p, 0) = \sum_i \theta_i p \cdot b_i \le \max\{0, \max_i p \cdot b_i\}$$
$$\le \bar{H}_2^{\mathrm{TI}}(x, z, p, 0).$$
(110)

Combining (109) and (110), we have

$$\bar{H}_W^{\text{TI}}(x, z, p, 0) = \bar{H}_2^{\text{TI}}(x, z, p, 0).$$
(111)

ii) Case 2: q < 0.

$$\bar{H}_2^{\mathrm{TI}}(x, z, p, q) = -q \max\left\{0, H\left(x, -\frac{p}{q}\right)\right\}$$
(112)

and, by the convexity of  $H_2^{\text{TI}}$  in  $p, H_2^{\text{TI}**} \equiv H_2^{\text{TI}}$ . Then

$$\bar{H}_W^{\mathrm{TI}}(x,z,p,q) = -qH_2^{\mathrm{TI}}\left(x,-\frac{p}{q}\right). \tag{113}$$

By combining (112), (113), and (21), we conclude the proof.

#### REFERENCES

- D. Lee and C. J. Tomlin, "Hamilton-Jacobi equations for two classes of state-constrained zero-sum games," 2021, arXiv:2106.15006.
- [2] D. Lee, A. Keimer, A. M. Bayen, and C. J. Tomlin, "Hamilton-Jacobi formulation for state-constrained optimal control and zero-sum game problems," in *Proc. IEEE 59th Conf. Decis. Control*, 2020, pp. 1078–1085.

- [3] J.-P. Aubin and H. Frankowska, "The viability kernel algorithm for computing value functions of infinite horizon optimal control problems," J. Math. Anal. Appl., vol. 201, no. 2, pp. 555–576, 1996.
- [4] C. J. Tomlin, J. Lygeros, and S. S. Sastry, "A game theoretic approach to controller design for hybrid systems," *Proc. IEEE*, vol. 88, no. 7, pp. 949–970, Jul. 2000.
- [5] J. Lygeros, C. Tomlin, and S. Sastry, "Controllers for reachability specifications for hybrid systems," *Automatica*, vol. 35, no. 3, pp. 349–370, 1999.
- [6] S. Bansal, M. Chen, S. Herbert, and C. J. Tomlin, "Hamilton-Jacobi reachability: A brief overview and recent advances," in *Proc. IEEE 56th Annu. Conf. Decis. Control*, 2017, pp. 2242–2253.
- [7] J. Lygeros, "On reachability and minimum cost optimal control," Automatica, vol. 40, no. 6, pp. 917–927, 2004.
- [8] K. Margellos and J. Lygeros, "Hamilton-Jacobi formulation for reachavoid differential games," *IEEE Trans. Autom. Control*, vol. 56, no. 8, pp. 1849–1861, Aug. 2011.
- [9] E. Barron and H. Ishii, "The Bellman equation for minimizing the maximum cost," *Nonlinear Anal., Theory, Methods, Appl.*, vol. 13, no. 9, pp. 1067–1090, 1989.
- [10] S. Di Marco and R. Gonzalez, "Relaxation of minimax optimal control problems with infinite horizon," J. Optim. Theory Appl., vol. 101, no. 2, pp. 285–306, 1999.
- [11] I. J. Fialho and T. T. Georgiou, "Worst case analysis of nonlinear systems," *IEEE Trans. Autom. Control*, vol. 44, no. 6, pp. 1180–1196, Jun. 1999.
- [12] H. M. Soner, "Optimal control with state-space constraint I," SIAM J. Control Optim., vol. 24, no. 3, pp. 552–561, 1986.
- [13] H. M. Soner, "Optimal control with state-space constraint II," SIAM J. Control Optim., vol. 24, no. 6, pp. 1110–1122, 1986.
- [14] H. Ishii and S. Koike, "A new formulation of state constraint problems for first-order PDEs," *SIAM J. Control Optim.*, vol. 34, no. 2, pp. 554–571, 1996.
- [15] H. Frankowska and R. Vinter, "Existence of neighboring feasible trajectories: Applications to dynamic programming for state-constrained optimal control problems," *J. Optim. Theory Appl.*, vol. 104, no. 1, pp. 20–40, 2000.
- [16] I. Capuzzo-Dolcetta and P.-L. Lions, "Hamilton-Jacobi equations with state constraints," *Trans. Amer. Math. Soc.*, vol. 318, no. 2, pp. 643–683, 1990.
- [17] H. Frankowska and S. Plaskacz, "Semicontinuous solutions of Hamilton– Jacobi-Bellman equations with degenerate state constraints," J. Math. Anal. Appl., vol. 251, no. 2, pp. 818–838, 2000.
- [18] A. Altarovici, O. Bokanowski, and H. Zidani, "A general Hamilton-Jacobi framework for non-linear state-constrained control problems," *ESAIM: Control*, Optim., Calculus Variations, vol. 19, no. 2, pp. 337–357, 2013.
- [19] O. Bokanowski, A. Picarelli, and H. Zidani, "State-constrained stochastic optimal control problems via reachability approach," *SIAM J. Control Optim.*, vol. 54, no. 5, pp. 2568–2593, 2016.
- [20] O. Bokanowski, A. Désilles, and H. Zidani, "Relationship between maximum principle and dynamic programming in presence of intermediate and final state constraints," *ESAIM, Control, Optim., Calculus Variations*, vol. 27, 2021, Art. no. 91.
- [21] N. Gammoudi and H. Zidani, "A differential game control problem with state constraints," *Math. Control Related Fields*, vol. 13, no. 2, pp. 554–582, 2023.
  [22] C. G. Claudel and A. M. Bayen, "Lax–Hopf based incorporation of
- [22] C. G. Claudel and A. M. Bayen, "Lax–Hopf based incorporation of internal boundary conditions into Hamilton-Jacobi equation. Part I: Theory," *IEEE Trans. Autom. Control*, vol. 55, no. 5, pp. 1142–1157, May 2010.
- [23] S. Osher and R. Fedkiw, Level Set Methods and Dynamic Implicit Surfaces, vol. 153. Berlin, Germany: Springer, 2003.
- [24] M. Korda, D. Henrion, and C. N. Jones, "Convex computation of the maximum controlled invariant set for polynomial control systems," *SIAM J. Control Optim.*, vol. 52, no. 5, pp. 2944–2969, 2014.
- [25] A. Oustry, M. Tacchi, and D. Henrion, "Inner approximations of the maximal positively invariant set for polynomial dynamical systems," *IEEE Control Syst. Lett.*, vol. 3, no. 3, pp. 733–738, Jul. 2019.
- [26] S. V. Rakovic and M. Baric, "Parameterized robust control invariant sets for linear systems: Theoretical advances and computational remarks," *IEEE Trans. Autom. Control*, vol. 55, no. 7, pp. 1599–1614, Jul. 2010.
- [27] S. Yu, C. Maier, H. Chen, and F. Allgöwer, "Tube MPC scheme based on robust control invariant set with application to Lipschitz nonlinear systems," *Syst. Control Lett.*, vol. 62, no. 2, pp. 194–200, 2013.
- [28] A. D. Ames, X. Xu, J. W. Grizzle, and P. Tabuada, "Control barrier function based quadratic programs for safety critical systems," *IEEE Trans. Autom. Control*, vol. 62, no. 8, pp. 3861–3876, Aug. 2017.

6495

- [29] J. J. Choi, D. Lee, K. Sreenath, C. J. Tomlin, and S. L. Herbert, "Robust control barrier–value functions for safety-critical control," in *Proc. IEEE* 60th Conf. Decis. Control, 2021, pp. 6814–6821.
- [30] L. Wang, A. D. Ames, and M. Egerstedt, "Safety barrier certificates for collisions-free multirobot systems," *IEEE Trans. Robot.*, vol. 33, no. 3, pp. 661–674, Jun. 2017.
- [31] E. Squires, P. Pierpaoli, and M. Egerstedt, "Constructive barrier certificates with applications to fixed-wing aircraft collision avoidance," in *Proc. IEEE Conf. Control Technol. Appl.*, 2018, pp. 1656–1661.
- [32] M. Jankovic, "Robust control barrier functions for constrained stabilization of nonlinear systems," *Automatica*, vol. 96, pp. 359–367, 2018.
- [33] S. V. Rakovic, E. C. Kerrigan, K. I. Kouramas, and D. Q. Mayne, "Invariant approximations of the minimal robust positively invariant set," *IEEE Trans. Autom. control*, vol. 50, no. 3, pp. 406–410, Mar. 2005.
- [34] I. M. Mitchell, J. Budzis, and A. Bolyachevets, "Invariant, viability and discriminating kernel under-approximation via zonotope scaling," in *Proc. 22nd ACM Int. Conf. Hybrid Syst., Computation Control*, 2019, pp. 268–269.
- [35] A. M. Carvalho and J. S. Sequeira, "Approximating viability kernels of high-dimensional linear systems," in *Proc. IEEE 18th Eur. Control Conf.*, 2019, pp. 4289–4294.
- [36] M. Rungger and P. Tabuada, "Computing robust controlled invariant sets of linear systems," *IEEE Trans. Autom. Control*, vol. 62, no. 7, pp. 3665–3670, Jul. 2017.
- [37] T. B. Blanco, M. Cannon, and B. De Moor, "On efficient computation of low-complexity controlled invariant sets for uncertain linear systems," *Int. J. Control*, vol. 83, no. 7, pp. 1339–1346, 2010.
- [38] L. Lindemann and D. V. Dimarogonas, "Control barrier functions for signal temporal logic tasks," *IEEE Control Syst. Lett.*, vol. 3, no. 1, pp. 96–101, Jan. 2019.
- [39] G. E. Fainekos, S. G. Loizou, and G. J. Pappas, "Translating temporal logic to controller specifications," in *Proc. IEEE 45th Conf. Decis. Control*, 2006, pp. 899–904.
- [40] Y. Chen and N. Ozay, "Data-driven computation of robust control invariant sets with concurrent model selection," *IEEE Trans. Control Syst. Technol.*, vol. 30, no. 2, pp. 495–506, Mar. 2022.
- [41] A. Taylor, A. Singletary, Y. Yue, and A. Ames, "Learning for safety-critical control with control barrier functions," in *Proc. Learn. Dyn. Control*, 2020, pp. 708–717.
- [42] A. Robey et al., "Learning control barrier functions from expert demonstrations," in Proc. IEEE 59th Conf. Decis. Control, 2020, pp. 3717–3724.
- [43] J. F. Fisac, N. F. Lugovoy, V. Rubies-Royo, S. Ghosh, and C. J. Tomlin, "Bridging Hamilton-Jacobi safety analysis and reinforcement learning," in *Proc. Int. Conf. Robot. Automat.*, 2019, pp. 8550–8556.
- [44] M. Bardi and L. C. Evans, "On Hopf's formulas for solutions of Hamilton-Jacobi equations," *Nonlinear Anal., Theory Methods Appl.*, vol. 8, no. 11, pp. 1373–1381, 1984.
- [45] P. D. Lax, "Hyperbolic systems of conservation laws II," Commun. Pure Appl. Math., vol. 10, no. 4, pp. 537–566, 1957.
- [46] E. Hopf, "Generalized solutions of non-linear equations of first order," J. Math. Mechanics, vol. 14, no. 6, pp. 951–973, 1965.
- [47] L. C. Evans, Partial Differential Equations. Providence, RI, USA: Amer. Math. Soc., 2010.
- [48] J. Darbon and S. Osher, "Algorithms for overcoming the curse of dimensionality for certain Hamilton–Jacobi equations arising in control theory and elsewhere," *Res. Math. Sci.*, vol. 3, no. 1, 2016, Art. no. 19.
- [49] I. Yegorov and P. M. Dower, "Perspectives on characteristics based curse-of-dimensionality-free numerical approaches for solving Hamilton– Jacobi equations," *Appl. Math. Optim.*, vol. 83, no. 1, pp. 1–49, 2021.
- [50] Y. T. Chow, J. Darbon, S. Osher, and W. Yin, "Algorithm for overcoming the curse of dimensionality for state-dependent Hamilton–Jacobi equations," *J. Comput. Phys.*, vol. 387, pp. 376–409, 2019.
- [51] D. Lee, S. A. Deka, and C. J. Tomlin, "Convexifying state-constrained optimal control problems," *IEEE Trans. Autom. Control*, early access, Nov. 11, 2022, doi: 10.1109/TAC.2022.3221704.
- [52] H. G. Bock and K.-J. Plitt, "A multiple shooting algorithm for direct solution of optimal control problems," *IFAC Proc. Vol.*, vol. 17, no. 2, pp. 1603–1608, 1984.
- [53] M. Kelly, "An introduction to trajectory optimization: How to do your own direct collocation," *SIAM Rev.*, vol. 59, no. 4, pp. 849–904, 2017.
- [54] S. Boyd and L. Vandenberghe, *Convex Optimization*. Cambridge, U.K.: Cambridge Univ. Press, 2004.

- [55] J. Nocedal and S. J. Wright, *Numerical Optimization*. New York, NY, USA: Springer, 1999.
- [56] T. Goldstein and S. Osher, "The split Bregman method for L1-regularized problems," *SIAM J. Imag. Sci.*, vol. 2, no. 2, pp. 323–343, 2009.
- [57] P.-L. Lions and J.-C. Rochet, "Hopf formula and multitime Hamilton-Jacobi equations," *Proc. Amer. Math. Soc.*, vol. 96, no. 1, pp. 79–84, 1986.
- [58] D. Lee and C. J. Tomlin, "A Hopf-Lax formula in Hamilton-Jacobi analysis of reach-avoid problems," *IEEE Control Syst. Lett.*, vol. 5, no. 3, pp. 1055–1060, Jul. 2021.
- [59] J. A. Sethian, "A fast marching level set method for monotonically advancing fronts," *Proc. Nat. Acad. Sci. USA*, vol. 93, no. 4, pp. 1591–1595, 1996.
- [60] J. C. Butcher and G. Wanner, "Runge-Kutta methods: Some historical notes," *Appl. Numer. Math.*, vol. 22, no. 1–3, pp. 113–151, 1996.
- [61] J. F. Bonnans and A. Festa, "Error estimates for the Euler discretization of an optimal control problem with first-order state constraints," *SIAM J. Numer. Anal.*, vol. 55, no. 2, pp. 445–471, 2017.
- [62] W. Alt, U. Felgenhauer, and M. Seydenschwanz, "Euler discretization for a class of nonlinear optimal control problems with control appearing linearly," *Comput. Optim. Appl.*, vol. 69, no. 3, pp. 825–856, 2018.
- [63] A. Domahidi, A. U. Zgraggen, M. N. Zeilinger, M. Morari, and C. N. Jones, "Efficient interior point methods for multistage problems arising in receding horizon control," in *Proc. IEEE 51st Conf. Decis. Control*, 2012, pp. 668–674.
- [64] J. Guanetti, Y. Kim, and F. Borrelli, "Control of connected and automated vehicles: State of the art and future challenges," *Annu. Rev. Control*, vol. 45, pp. 18–40, 2018.



**Donggun Lee** received the B.S. and M.S. degrees in mechanical engineering from the Korea Advanced Institute of Science and Technology (KAIST), Daejeon, South Korea, in 2009 and 2011, respectively. He is currently working toward the Ph.D. degree in mechanical engineering with the University of California, Berkeley, CA, USA.

His research interests include control and game theory with applications to robotics, autonomous driving, multiagent systems, and out-

door robot navigation.



**Claire J. Tomlin** (Fellow, IEEE) received the Ph.D. degree in electrical engineering and computer sciences from the University of California, Berkeley, CA, USA in 1998. She is currently the James and Katherine Lau Professor of Engineering with EECS, University of California, Berkeley, Berkeley, CA, USA. She was an Assistant, Associate, and Full Professor in aeronautics and astronautics with Stanford University from 1998 to 2007, and in 2005, she joined UC Berkeley. She works in the area of control theory

and hybrid systems, with applications to air traffic management, UAV systems, energy, robotics, and systems biology.

Prof. Tomlin is a MacArthur Foundation Fellow in 2006, was awarded the IEEE Transportation Technologies Award in 2017, and was elected to the National Academy of Engineering and the American Academy of Arts and Sciences in 2019.